

Live/Work

8'-0" high windows Incredible visibility and exposure 2 story units Build out and Finish options Perfect walkable location







- 1: Selection of desired unit
- 2: Review proposed use and compliance with Zoning requirements See below
- 3: Complete reservation form
- 4: Review Condominium Documents
- 5: Review interior layout and any modifications
- 6: Review finishes and any modifications
- 7: Review signage needs
- 8: Assess potential cost impacts (positive or negative)
- 9: Complete Offer to purchase

Applicable Code Sections:

Section 419 of the International Building Code

- City of Monona Zoning Sections:
- 480-27 Neighborhood Business District
- 480-28 Retail Business District, only including:

"service, office, financial, professional, ancillary residential and entertainment activities serving the community"

Prohibited Uses:

Storage and/or use of hazardous materials Adult Entertainment Repair Shops Production or Manufacturing facilities Warehousing or Distribution Centers Full Service Dine-in Restaurants











































































Door Color

Natural- Option "A" Pecan- Option "B"



Cabinet Pulls








Solid Surface Countertops





Backsplash

Backsplash



Single-lever sink mixer 1/2" MODEL # 30295000

Minnesota Snow

Pure Freude an Wasser

GROHE America, Inc | 200 North Gary Avenue, Suite G, Roselle, IL 60172 Phone: +1 (800) 444-7643 | Fax: +1 (800) 225-2778 | us-customerservice@grohe.cor





Product Description: Single-lever sink mixer 1/2"
Standard Specification:
 Single hole installation
 GROHE StarLight finish
 GROHE SilkMove 28 mm ceramic cartridge
 Professional spray
· Locking Push Button Control - to switch from regular
to spray
 Swivel spout
 Swivel area 360°
 Protected against backflow
 Stainless steel flex lines
 Quick installation system
 Automatic return to aerator
 Max Flow Rate 1.75 gpm

Color: a 30295000 000 chrome







Portas Savona Hex



















BOSCH 800 Series

20.5 Cu. Ft. Stainless Steel Refrigerator30" Brushed Stainless Steel with Black Glass Range1.2 Cu. Ft. Stainless Steel Over The Range Microwave24" Stainless Steel Dishwasher







HOME APPLIANCES

Single Unit LG WashTower™ with Center Control™ 4.5 cu. ft. Front Load Washer and 7.4 cu. ft. Dryer

Ultra Large Capacity 4.5 cu.ft. Washer and 7.4 cu.ft. Dryer Single Unit WashTower™ Design Built-In Intelligence - AI Fabric Sensors/Smart Learner/Smart Pairing™

Advanced Washing & Drying - TurboWash™ 360°/Allergiene™ wash cycle/ TurboSteam™

ThinQ[®] Technology / Proactive Customer Care Tempered Glass Door

LG ThinQ

KEY FEATURE

CEF

Enjoy new ways to control your home with LG smart appliances Use the TimiO* app to start the landry while you're out Get alerts when the cycle is finished With Smart Parinog", the washer can tell the dryer to select a compatible drying cycle



Drying Levels		Very, More, Norn
		Less, Damp
Manual Dry Times		60 min., 50 min., min., 30 min., 20 r
FABRIC CARE FEATURES	Washer	Dryer
AI DD	Yes	-
TurboWash™ 360 Technology	Yes	
Steam Technology	Yes	
Allergiene™ Cycle	Yes	
Sanitary Cycle	Yes	
SenseClean [™] System	Yes	
Al Dry		Yes
Sensor Dry		Yes
Precise Temperature Control with Variable Heater		Yes
Steam Technology		Yes (TurboSteam
CONVENIENCE FEATURES	Washer	Dryer
TrueBalance [™] Anti-Vibration System	Yes	
	Prewash, Main	
	Wash (with liquid	
4 Tray Dispenser	detergent cup),	
	Bleach, Fabric	
	Softener	
LoDecibel [™] Quiet Operation	Yes	
End of Cycle Signal	Yes	
LoadSense	Yes	-
Child Lock	Yes	
Auto Suds Removal	Yes	-
Leveling Legs	4 Adjustable Legs	
Easy Loading TilTub™	Yes	
Drum Light	Yes	-
Remaining Time Display/Status Indicator(s)	Yes	
3 Minute Installation Check		Yes
LoDecibel [™] Quiet Operation		Yes
End of Cycle Signal		Yes
Drum Light		Yes
FlowSense™ Duct Clogging Indicator		Yes
Wrinkle Care Option		Yes
Child Lock		Yes
Venting Option		Electric: 3 Way Venting / Gas: 2 V Venting
4 Adjustable Legs		Yes
Remaining Time Display/Status		Yes





AVAILABLE COLORS

BLACK STEEL

WHITE

LG Life's Good

SmartDiagnosis™ (v3.0))	fes
Wi-Fi)	/es
Remote Start and Cycle Monitor)	(es
Energy Monitoring	1	/es
Tub Clean Coach	Yes	-
MOTOR AND AGITATOR	Washer	Dryer
Motor Type	Inverter Direct Drive Motor	
Axis	Horizontal	
MATERIALS AND FINISHES	Washer	Dryer
NeveRust [™] Stainless Steel Drum	Yes	-
Cabinet	PCM	
Control Panel	Plastic	
Top Plate	Painted	-
NeveRust [™] Stainless Steel Drum		Yes
Tub Rear		Painted
CERTIFICATION	Washer	Dryer
AAFA	Yes	
POWER SOURCE	Washer	Dryer
Ratings	UL Listed	-
Electrical Requirements	120V, 10 Amps	-
Туре	Electric	-
Ratings	-	CSA Listed
Electrical Requirements	-	120V, 15 Amps (Ga 240V 30 Amps (Electric)
Туре		Gas / Electric
BTU Rating		10,000 Feet
LP Conversion Kit		383EEL3002D
Side Venting Kit	-	383EEL9001B
DIMENSIONS	Washer	Dryer
Product (WxHxD)	27° × 74 3	/8" x 30 3/8"
Depth with Door Open	55" D with	h door open
Carton (WxHxD)	30 1/16* x 7	9 11/32" x 32"
Weight (Product/Carton)	Electric: (326 lb / 361 lb) Gas: (329 lb / 364 lb	
LIMITED WARRANTY	Washer	Dryer
Parts and Labor	1 Year	1 Year
Direct Drive Motor	10 Years	-
Drum	3 Years	3 Years
UPC CODES		
WKEX200HBA (Black Steel)	048231028486	
WKGX201HBA (Black Steel)	048231029254	
WKEX200HWA (White)	048231028462	
WKGX201HWA (White)	048231029261	

© 2020 LG Electronics USA, Inc. All rights reserved. "LG Life's Good" is a registered trademark of LG Corp. All other product and brand names are trademarks or registered trademarks of their respective owners. Design, features and specifications are subject to change without notice. Non-metric weights and measurements are approximate. Some features may require internet access. Content and services may vary by product and are subject to change without notice. 10/19/20





FEATURES

- Individual Private Entrances
- Shared open spaces
- High Visibility Location
- Signage options
- Efficient, Optimized, and Durable Construction
- Low Utility costs
- Natural Light
- •Flexible and Customizable Spaces
- Innovative and Forward Thinking environment
- Designed to maximize Community interaction, walkability, and foot traffic

Option A (2)

	Flooring:	Туре	Manuf.	Style	Color		Size	Installation	Notes	Option
F1	Entry			Porcelian	White					
F1	Bathroom			Porcelian	White					
F2	Living Room	LVP		LVP	Pebble		9x72	Glue Down		
F2	Kitchen	LVP		LVP	Pebble		9x72	Glue Down		
F3	Bedroom	LVP					9x72	Glue Down		
T 1	Transistions	Aluminum	Schluter	Vinpro-S	Brushed Nickel	VPSL70ATGB				Prefer black if available
	Kitchen	Туре	Manuf.	Style	Color			Notes		
C1	Base Cabinets				Fawn					
C1	Wall Cabinets				Fawn					
CT1	Countertop	Quartz	Cambria		Minnesota Sno	w				
BS1	Backsplash				Black.white					
					DIGCK.WHILE					
P1	Pulls				Satin Nickel					
W1 W2 W3	Pulls Walls Bathroom Living Room Kitchen Bedroom	Material GB GB GB GB		Finish Paint Paint Paint Paint		Base Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	Size .9"x .3" .9"x .3" .9"x .3" .9"x .3"	Adhesive Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den	GB GB GB GB GB		Paint Paint Paint Paint Paint	Satin Nickel Color TBD TBD TBD TBD TBD TBD TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"x .3" .9"x .3" .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors	GB GB GB GB GB Material	Frame	Paint Paint Paint Paint Paint Style	Satin Nickel Color TBD TBD TBD TBD TBD TBD TBD Color	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry	GB GB GB GB GB Material Wood	K/D	Paint Paint Paint Paint Paint Style Flush	Color TBD TBD TBD TBD TBD TBD TBD Color TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom	GB GB GB GB Material Wood Wood	K/D K/D	Paint Paint Paint Paint Paint Style Flush Flush	Satin Nickel Color TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet	GB GB GB GB Material Wood Wood Laminate	K/D K/D Murphy Hardware	Paint Paint Paint Paint Paint Style Flush Flush See A8.02	Satin Nickel Color TBD TBD TBD TBD TBD TBD TBD TBD TBD TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet Bedroom	GB GB GB GB Material Wood Wood Laminate Wood/Laminate	K/D K/D Murphy Hardware Slider/Hinge	Paint Paint Paint Paint Paint Flush Flush See A8.02 Flush	Satin Nickel Color TBD TBD TBD TBD TBD Color TBD TBD TBD TBD TBD TBD TBD TBD TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet	GB GB GB GB Material Wood Wood Laminate	K/D K/D Murphy Hardware Slider/Hinge	Paint Paint Paint Paint Paint Style Flush Flush See A8.02	Satin Nickel Color TBD TBD TBD TBD TBD TBD TBD TBD TBD TBD	Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner Fasade Large Corner	.9"× .3" .9"× .3" .9"× .3" .9"× .3" .9"× .3"	Adhesive Adhesive Adhesive	Prefinished Prefinished Prefinished	Quarter round or shoe Quarter round or shoe Quarter round or shoe

Option B

_

	Flooring:	Туре	Manuf.	Style	Color		Size	Installation	Notes	Option
F1	Entry	21		Porcelain	Grey		Mosaic			•
F1	Bathroom			Porcelain	Grey		Mosaic			
F2	Living Room	LVP		LVP	Pebble		9x72	Glue Down		
F2	Kitchen	LVP		LVP	Pebble		9x72	Glue Down		
F3	Bedroom	LVP		LVP	Pebble		9x72	Glue Down		
T1	Transistions	Aluminum	Schluter	Vinpro-S	Brushed Nicke	e VPSL70ATGB				
	Kitchen	Туре	Manuf.	Style	Color			Notes		
C1	Base Cabine				Natural					
C1	Wall Cabinet				Natural					
CT1	Countertop	Quartz	Cambria		Fieldstone					
BS1	Backsplash	Quant	oambila		White/charco	bal				
P1	Pulls			BLACK OR	Pewter Antiqu	эс				
	Pulls Walls	Material		BLACK OR	Pewter Antiqu	Je Base	Size			
P1		Material Porcelain					Size			
	Walls Bathroom	Porcelain		Finish	Color			Adhesive	Prefinished	Quarter round or sho
	Walls Bathroom	Porcelain		Finish Tile	Color Pattern	Base	i∈ .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	
W1 W2	Walls Bathroom Living Room	Porcelain GB		Finish Tile Paint	Color Pattern TBD	Base Fasade Larg	je.9"x.3" je.9"x.3"			Quarter round or sho
W1 W2 W3	<u>Walls</u> Bathroom Living Room Kitchen	Porcelain GB GB		Finish Tile Paint Paint	Color Pattern TBD TBD	Base Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive	Prefinished	Quarter round or sho Quarter round or sho Quarter round or sho Quarter round or sho
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors	Porcelain GB GB GB GB Material	Frame	Finish Tile Paint Paint Paint Paint Style	Color Pattern TBD TBD TBD TBD TBD Color	Base Fasade Larg Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or sho Quarter round or sho
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry	Porcelain GB GB GB GB Material Wood	K/D	Finish Tile Paint Paint Paint Paint Style Flush	Color Pattern TBD TBD TBD TBD TBD Color TBD	Base Fasade Larg Fasade Larg Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom	Porcelain GB GB GB GB Material Wood Wood	K/D K/D	Finish Tile Paint Paint Paint Paint Style Flush Flush	Color Pattern TBD TBD TBD TBD TBD TBD TBD TBD	Base Fasade Larg Fasade Larg Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet	Porcelain GB GB GB Material Wood Wood Laminate	K/D K/D Murphy Hard	Finish Tile Paint Paint Paint Paint Style Flush Flush V See A8.02	Color Pattern TBD TBD TBD TBD TBD TBD TBD TBD TBD	Base Fasade Larg Fasade Larg Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or sho Quarter round or sho
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet Bedroom	Porcelain GB GB GB Material Wood Laminate Wood/Lamin	K/D K/D Murphy Hard n: Slider/Hinge	Finish Tile Paint Paint Paint Paint Style Flush Flush V See A8.02 Flush	Color Pattern TBD TBD TBD TBD TBD TBD TBD TBD TBD TBD	Base Fasade Larg Fasade Larg Fasade Larg Fasade Larg	le .9"x .3" le .9"x .3" le .9"x .3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or shoe Quarter round or shoe
W1 W2 W3 W4	Walls Bathroom Living Room Kitchen Bedroom Den Doors Entry Bathroom Utility Closet	Porcelain GB GB GB Material Wood Laminate Wood/Lamin	K/D K/D Murphy Hard	Finish Tile Paint Paint Paint Paint Style Flush Flush V See A8.02 Flush Flush	Color Pattern TBD TBD TBD TBD TBD TBD TBD TBD TBD	Base Fasade Larg Fasade Larg Fasade Larg Fasade Larg	ie.9"x.3" ie.9"x.3" ie.9"x.3" ie.9"x.3" ie.9"x.3"	Adhesive Adhesive	Prefinished Prefinished	Quarter round or shoe Quarter round or shoe



CITY OF MONONA SIGN PERMIT APPLICATION FORM

For all sign standards and requirements see Monona Municipal Code <u>Chapter 480, Article XII</u> and <u>Attachment</u> <u>5 – Sign Groups</u>. Contact the Planning Department with any questions at (608)222-2525 or <u>planning@ci.monona.wi.us</u>

APPLICANT INFORMATION*:	
Address of Sign to	
be Located:	
Applicant Name	Business
Applicant Name:	Name:
Email:	Phone:
Applicant Signature:	Date:
PROPERTY OWNER INFORMATION:	
Property Owner	
Name:	
Property Owner Signature ⁺ :	Date:
SIGN CONTRACTOR INFORMATION:	
Sign Contractor	Company
Name:	Company:
Address:	
Email:	Phone:

*A sign permit application may be granted to any tenant or lessee acting as agent of the owner with written consent of the owner of the building, structure, and land on which the sign is to be erected. However, such applicant shall be held responsible and liable to prove his/her right for such a permit if contested by any aggrieved party.

[†]Applicants may also submit a letter or email demonstrating approval from the property owner

Below Space for Office Use				
Date Approved:	Permit Number:			
Approved By:	Fee:	Paid:		
Attach conditions for approved permits.				

SIGNAGE DETAILS

Number of Signs:							
Types of Sign (check all that apply):							
□Landscape Ground	□Pylon	□Wall	□Off Premises	Directional	□Reader Board		
□Other							
Sign Materials (check a	ll that apply):						
□Wood		□Aluminum	□Masonry	□Plastic	□Vinyl		
□Other							
Lighting types (check al	ll that apply):						
Externally Illuminated Internally Illuminated Non-Illuminated							
Sign Dimensions							
Height of Freestanding	Sign (From Grad	le to Top):					
Total Square Feet of all	Total Square Feet of all signs (One Side Face):						

SIGN PERMIT APPLICATION PROCEDURES

Applications must include:

- □ Completed sign permit application form
- A scale drawing of the proposed sign(s) showing the message to be displayed, sign area, height, material composition, colors, typestyle and size, all graphic elements to be used, and landscaping coordinated with site landscaping for landscape ground signs
- □ A photograph showing the location of the proposed sign and its relationship to the building or surrounding area
- \Box A dimensioned plot plan drawn to scale (1" = 20') showing the location of the lot, building or structure on which the proposed sign is to be attached or erected
- □ A description of all electrical equipment and attachments if the sign is to be lighted or illuminated (illumined signs require an electrical permit from the building inspector)

Other requirements:

- □ Applicants requiring Plan Commission review shall submit 10 hard copy sets of the complete application.
- □ The applicant or sign contractor shall be present at the Plan Commission meeting or the sign application may not be considered.
- Submit electronic (PDF) files of all applications via email to the City Planner at planning@ci.monona.wi.us
- □ All freestanding signs 20 feet tall or grater shall submit construction plans stamped by a certified engineer to the City Building Inspector for approval.
- □ Shopping centers / multi-tenant buildings with leased space shall develop a comprehensive signage plan.
- □ Temporary sign applications shall include a written statement of the dates on which the sign is to be displayed.
- □ The fee shall be one dollar (\$1.00) per square foot for all signs with a minimum fee of \$50.00 and is due before issuance of any sign permit by the Zoning Administrator.

ARTICLE XII Signs, Canopies, Awnings and Billboards [No. 9-17-689; 5-20-2019 by Ord. No. 5-19-717]

§ 480-62. Definition of "sign".

- A. Definition of a sign. In this article, the word "sign" means any object, device, display, structure, or part thereof, situated or visible from outdoors, which is used to advertise, identify, display, direct or attract attention to an object, person, institution, organization, business, product, service, event, or location by any means, including words, letters, figures, designs, logos, symbols, fixtures, colors, illumination, or projected images.
 - (1) On-building sign category. A type of sign permanently affixed to an outside wall of a building and shall be made of durable all-weather materials. There are the following on-building sign types:
 - (a) Wall sign. A type of on-building sign that is mounted directly on, and parallel to, a building facade or other vertical building surface. A wall sign also includes a sign located on the interior of a building that is intended to be viewed primarily from beyond the boundaries of the site. Whether an interior sign is considered a wall sign shall be determined by the Zoning Administrator during the sign permit review process.
 - 1. The top edge of a wall sign shall not extend above the top edge of the vertical exterior wall or above the lowest edge of a roofline of the portion of the building to which it is mounted.
 - 2. <u>Wall signs shall not project more than 18 inches horizontally</u> beyond the face of any wall or other surface to which they are mounted.
 - (b) Projecting sign. A type of on-building sign that is mounted at any angle other than parallel to the wall on which it is mounted, extending from the face of the wall.
 - (3) Pedestrian category. A sign attached perpendicularly to the face of a building and mounted above sidewalk level, and which is oriented and sized for visibility to nearby pedestrians rather than to motorists. See Table 480-70(1) for exact dimensional requirements of this category. The sign shall be made of durable all-weather materials. The following pedestrian sign types are regulated by this article:
 - (a) Blade sign. A type of pedestrian sign that is mounted perpendicular to the wall on which it is mounted, extends less than four feet from the wall.
 - (b) Suspended sign. A type of pedestrian sign that is mounted perpendicular to the nearest wall and suspended from the underside of a horizontal plane surface, such as a covered porch, arcade, or canopy.
 - (4) Daily notice sign category. A sign typically used to advertise daily specials, daily menu items, or on-site events that change on a daily basis, which usually includes changeable

copy. This type of sign is often associated with restaurants, taverns, retail stores, and music venues. There are the following daily notice sign types:

- (a) Menu board sign. A type of daily notice sign mounted flat against a wall containing changeable copy.
 - 1. Menu board signs shall be securely affixed to the exterior wall of the building containing the use.
 - 2. Menu board signs shall not extend more than four inches from the wall on which they are mounted.
- (1) Window sign category. A sign located within a building that is attached to the inside face of an exterior window. The following window sign types are addressed by this article:
 - (a) Window sign. A type of sign that is either painted onto a window, attached to the inside face of an exterior window, or located inside a building within three feet of a window and intended to be viewed from the exterior of the building. Window signs may face toward the outside, the inside, or both. They shall not be placed on a door window, other windows needed to be clear for pedestrian safety, or on any window outside of the ground-floor.
- (2) Temporary board and banner sign category. A sign located outside of a building for a period not to exceed 28 calendar days in any six-month period. Such signs are often used for the purpose of informing the public of a sale or special offer. There are the following temporary board and banner sign types:
 - (a) Board sign. A type of temporary board and banner sign that is placed on the ground and is made of rigid material.
 - [1] Board signs are not permitted in the City of Monona.
 - (b) Sandwich board sign. A type of temporary board and banner sign placed on the ground and constructed in such a manner as to form an "A"-like shape, hinged or not hinged at the top. Each angular face shall be held together by a supporting element such as a folding bar, latch, or chain.
 - [1] Anything attached to a sandwich board sign shall not project outside the perimeter of the sign face nor project in excess of one inch from the sign face.
 - (c) Banner sign. A type of temporary board and banner sign that is made of flexible material such as cloth or vinyl and is supported along one or more sides or at two

or more corners by wires, ropes, string, nails, or other removable fastening materials.

- (d) Feather sign. A type of temporary board and banner sign consisting of a piece of vertically elongated, flexible material such as cloth or vinyl which is affixed to a single pole driven into the ground. The pole may be rigid or flexible but is not permanent.
 - [1] Feather signs are not permitted in the City of Monona.

PROHIBITED SIGNS

- (3) Flashing/scrolling/animated sign. A sign having lights or illumination which flashes, scrolls, moves, rotates, twinkles, blinks, flickers, varies in intensity of color, or uses intermittent electrical pulsations. Electronic message signs meeting the definition and requirements of § 480-67 of this article shall not be considered flashing, scrolling, or animated signs.
- (4) Floodlighted signs. Reflection illuminated signs whose light source is positioned so that 25% or more of light intensity directly from the light source is visible from a public right-of-way by vehicular traffic or whose light source is visible from residential property.
- (5) Inflatable sign. A sign capable of being filled with and expanded by air or other gas, including animated or "dancing" inflatable signs.
- (6) Mobile/portable sign. A sign mounted on a frame or chassis designed to be easily relocated, including unlicensed or inoperable vehicles and/or trailers. Licensed and operable business vehicles, trailers, or other pieces of equipment shall not be considered mobile or portable signs.
- (7) Off-premise advertising sign. A sign which directs attention to activity that is conducted upon the site where the sign is displayed. Off-premise advertising signs include billboards.
 - (a) Existing legal off-premise advertising signs made nonconforming by this section shall be permitted to continue as legal, nonconforming signs, subject to the requirements of § 480-94.
- (8) Roof sign. A sign displayed above the eaves or cornice of a building.
- B. Other definitions.
 - (1) Building frontage. The width of the building facade that fronts a public street, highway, or interstate.
 - (2) Business/tenant frontage. The portion of a building frontage occupied by a single tenant space having a public entrance within the building frontage. For businesses located on

the interior of a building without frontage, the building elevation providing customer access shall be considered the business frontage.

- (3) Copy. Words, letters, numbers, figures, designs, or other symbolic representations incorporated into a sign.
- (4) Changeable copy. Sign copy that may be changed manually to provide different information such as boards with changeable letters, bulletin boards, and chalkboards.
- (5) Customer entrance. The entrance that the public can use when an establishment is open to the public.
- (6) Electronic message sign. See § 480-67.
- (7) Elevation, building. The view of any building or other structure from any one of four sides regardless of the configuration or orientation of a building. No building shall be treated as having more than four building elevations. Each elevation will generally be identified as a north, south, east or west building elevation.
- (8) External illumination. The lighting of an object from a light source located a distance from the object.
- (9) Facade. See "elevation."
- (10) Height of sign. The vertical distance from the average ground level at the base of the sign to the top of the highest attached component of the sign. See § 480-74B for the measurement of sign height.
- (11) Maintain. Maintaining the existing appearance of the sign; replacing the sign face or the supporting structure with identical materials, colors, and messages; changing the message of a marquee sign; or changing the face of an off-premise advertising sign (billboard).
- (12) Lighting, ambient. Illumination in which the only light that falls onto the sign come from sources that are available naturally (e.g., sunlight, moonlight) or from artificial lighting sources used for other purposes in the vicinity of the sign (e.g., street lights, lighting installed for other purposes or sites)
- (13) Lighting, backlit. Illumination that is arranged in such a way that the light is cast from behind the sign to the eyes of the viewer. Often, the lighting element is unshielded but concealed behind individual freestanding letters, creating a silhouette effect.
- (14) Lighting, gooseneck. Illumination resulting from light emitted directly from a shielded light fixture located at the top of the sign and angled downward onto the sign face. The light fixture is attached to a curved neck which is often flexible, allowing the user to position the light source onto the sign face.
- (15) Lighting, internal. Illumination emanating from a lighting element that is located behind the sign face and which is completely enclosed.

- (a) Internally illuminated signs shall have individual channel letters or silhouettes, light-colored copy on a dark-colored or opaque cabinet field background (so the copy is legible during the day and night), and a translucent message. When illuminated, the sign shall appear to have an illuminated copy with a dark or nonilluminated background.
 - [1] The requirements of Subsection F(15)(a), above, shall not apply to internally illuminated individual characters, letters, or shapes that do not contain copy on the sign face.
- (b) Neon lighting is not considered to be internal lighting.
 - (16) Lighting, neon. Glass tube lighting in which a gas and phosphors are used in combination to create a colored light. Neon lighting is not considered internal lighting as defined in Subsection F(15) above.
 - (17) Sign area. The entire face of a sign, including the extreme limits of writing, representation, emblem, or other display, together with any material or color forming an integral part of the background of the display but not including any supporting framework. See § 480-74D.
 - (18) Plat phase. The collection of lots, rights-of-way, and outlots located within the perimeter boundary of a final plat and intended to be developed concurrently.
 - (19) Site. A site shall include all lots that are contiguous, under unified single ownership and intended to remain under unified single ownership under the jurisdiction of this article. A site may also be determined to be a portion of a single lot where multiple building are present or more than one building contains separate businesses (approved by the City). For the purposes of this article, the site shall be determined by the Zoning Administrator.
- (20) Sign face. The area or display surface used for the message.
 - (21)Temporary sign. A sign or advertising intended to be displayed for a certain limited period of time. If a sign display area is permanent, but the message displayed is subject to periodic changes, that sign shall not be considered as temporary. A mobile or portable sign shall not be considered a temporary sign or used for such a purpose. Refer to Figure 480-70(2) and Figure 480-70(4) for rules related to temporary signs.
 - (22) Three-dimensional signs. Signs that have a depth or relief on their surface. Where a sign consists of one or more three-dimensional objects (i.e., balls, cubes, clusters of objects, sculpture or statue-like trademarks), fitting within the smallest simple geometric shape (rectangle, circle, or triangle), the sign area shall be measured as their maximum projection upon a vertical plane. See § 480-74D(5).
 - (23) Window. The total area within a single window pane.

§ 480-70. Permitted sign rules.

- A. Signs shall be allowed on private property in the City in accordance with Figures 480.70(1) through 480.70(4), which address permitted signage as it relates to permits, quantity, area, location, lighting, and zoning districts. The requirements set forth in Figures 480.70(1) through 480.70(4) shall be declared to be part of this article.
 - (1) The rules for permanent business signs are located in Figure 480.70(1).
 - (2) The rules for temporary business signs are located in Figure 480.70(2).
 - (3) The rules for permanent miscellaneous signs are located in Figure 480.70(3).
 - (4) The rules for temporary miscellaneous signs are located in Figure 480.70(4).
 - (5)The rules for on-premise sign locations are located in Figure 480-70(5) and 480-70(6).
 - (6) Signage for all Planned Community Development projects shall be permitted per the base zoning district and may be granted flexibility through the Planned Community Development process.

§ 480-71. through § 480-73. (Reserved)

§ 480-74. Sign setback, height, measurement, and flexibility.

A. Sign setbacks. Signs must be located on-premises, outside of the vision triangle, and must not be located within any street right-of-way, unless otherwise specified or approved by the Plan Commission. Each sign type must be setback in a location consistent with Figure

480-70(1), Figure 480-70(2), Figure 480-70(3), Figure 480-70(4), and Figure 480-70(5).

- B. Sign height.
 - (1) The height of a freestanding sign shall be measured from the average ground level adjacent at the base of the sign to the top of the highest attached component of the sign, or from the centerline grade of the nearest adjacent public road, if such information is supplied with the permit application and confirmed by the Zoning Administrator, whichever is higher.
 - (2) The average ground level is defined as the average elevation within a three-foot radius of the sign on the ground upon which the sign supports are placed, except when the sign supports rest upon a berm or other area elevated above the surrounding ground. In such cases, the average elevation of the base of such berm or other area shall be considered as the ground level.
- C. Minimum ground clearance. All on-building and pedestrian signs shall have a minimum clearance of 14 feet above a drive, alley, street, parking space, or other vehicle surface and a minimum of eight feet above a sidewalk, patio, or other ground-level surface.
- D. Measurement of sign area: The measurement of sign area is based on the arrangement of sign

copy and sign background (sign background does not include building architectural elements such as painted strips or exterior building material changes, as determined by City staff):

- (1) In the case of a freestanding sign (including three-dimensional objects), sign area shall include the total sign area(s) that can be viewed from any single vantage point (i.e., for a typical freestanding sign that faces two directions, only the largest sign face visible from any single vantage point shall count toward the total permitted sign area).
- (2) For signs comprised of letters and related copy surrounded by one or more sign background colors on a single panel, frame, or cabinet, the sign area shall be measured as the smallest single rectangle enclosing the entire sign message and any and all background color areas. See Example 1 in Figure 480-74(2).
- (3) For signs comprised of individual letters and related copy surrounded by one or more sign background colors on individual panels, sign area shall be measured as the smallest single rectangle enclosing the entire sign message and any and all background color areas. See Example 1 in Figure 480-74(2).
- (4) Groupings of related smaller signs shall be counted as one sign. In such cases, sign area shall be measured as the smallest single rectangle enclosing the individual signs and any and all background color areas. See Example 2 in Figure 480-74(2).
- (5) For signs comprised of one or more three-dimensional objects (i.e., balls, cubes, clusters of objects, sculpture or statue-like trademarks), fitting within the smallest simple geometric shape (rectangle, circle, or triangle), the sign area shall be measured as their maximum projection upon a vertical plane. See Figure 480-74(1) below.



Figure 480-74(1): Measurement of 3D Sign Area

Figure 480-74(2): Measurement of Sign Area

Example 1: Individual letters or words grouped on one or more panels Smallest single rectangle



Example 2: Individual letters on separate panels Smallest single rectangle

§ 480-86. Sign permits - application, enforcement, and revocation.

- A. Applicability.
 - No freestanding sign listed under § 480-64(A)(1), on-building sign listed under § 480-64(A)(2), order board sign under § 480-64(A)(4)(b), on-site warning sign under § 480-64(C)(1)(c), on-site directional sign under § 480-64(C)(1)(d), or institutional information sign under § 480-64(C)(2)(a) shall be erected, installed, or constructed without approval from the City of Monona Plan Commission.
 - (2) No pedestrian sign under § 480-64(A)(3), menu board sign under § 480-64(A)(4)(a), window sign under § 480-64(B)(1), temporary board and banner sign under § 480-64(B)(2), or metal plaque sign or flag sign under § 480-64(C)(2) shall be erected, installed, or constructed without the granting of a permit from the Zoning Administrator in accordance with the provisions of this section.
 - (3) No pylon or marquee signs under §§ 480-64(A) and 480-64(B) shall be erected, installed, constructed, or maintained without the structural support being designed by a licensed professional engineer and approved by the Building Inspector as in compliance with the Building Code of the City. All frames and supports shall be of metal and designed to withstand pressure as provided in § 480-92.
 - (4) For signs requiring a permit in Subsection A(1), above, this section shall apply and be construed to require a permit for a change of copy on any sign or for any conversions or changes in the sign structure.
 - (5) This section shall not apply to repainting or re-facing with the same sign copy, cleaning, repair, or other normal maintenance of the sign or sign structure.
 - (6) No new permit is required for signs which are in place as of the effective date this article, and such signs may remain as legal nonconforming signs. Any alteration or

relocation of such signs shall conform to the requirements of this article. Refer to § 480-94 for rules pertaining to nonconforming signs.

- (7) Any sign permit granted hereunder shall not be assigned or transferred to any other sign, including a modified sign face or modified sign structure.
- (8) The owner or tenant may request all such signs at one site be included under one permit.
- B. Review authority.
 - (1) The Plan Commission shall review all freestanding, on-building, order board, on-site warning, on-site directional, or institutional information signs, all signs accompanying changes in use or new use which must receive approval of a zoning permit, and all proposed signs which would require approval of a comprehensive signage plan for a site or sites which will have more than one sign viewed together as part of a group of signs. This may be required with a zoning permit for change of use or a sign permit. All signs which require approval of a sign permit by the Zoning Administrator or Plan Commission shall be reviewed according to the following evaluation factors:
 - (a) Conformance to the zoning and sign code.
 - (b) Minimization of conflict with vehicular or pedestrian circulation.
 - (c) Compatibility with the building characteristics, adjacent uses, and adjacent signs.
 - (d) Compatibility with the specific physical site conditions which warrant approval of the proposed sign.
 - (e) Materials and maintenance aspects.
 - (f) Legibility and visual clarity.
 - (2) The Plan Commission may grant special exception to this article upon demonstration of due cause. Every applicant for a special exception to the sign district requirements shall submit a written statement to the Plan Commission which explains the reason for the request and how it meets the evaluation factors.
- C. Sign permit application. Each sign permit application shall include:
 - (1) The name, address, phone number, and email address of the applicant.
 - (2) The name of the business or land use the proposed sign will serve.
 - (3) The name, address, phone number, email address, and signature of the property owner.
 - (4) The name, address, phone number, and email address of the sign contractor.
 - (5) The property's zoning designation.
 - (6) The property's current land use or uses for the entire subject property, including all indoor and outdoor areas.
- (7) An aerial photograph or approved site plan for the subject property. If a site plan was Downloaded from https://ecode360.com/MO3595 on 2024-08-07

not previously required or approved, a site plan for the subject property with requirements as determined by the Zoning Administrator shall be provided. At a minimum, the site plan shall include the following:

- (a) Location, type, height, width, and area of the proposed sign.
- (b) Location, type, height, width, and area of all existing signs on the property and indication of whether existing sign(s) will remain or be removed/replaced.
- (c) For all freestanding and institutional information signs, the location and type of landscaping coordinated with site landscaping.
- (d) All property lines and buildings on the property and within 50 feet of the proposed sign.
- (e) All parking areas, driveways, and public roads.
- (f) Method of attachment, structural support, method of illumination, and sign materials.
- (g) The total area of all signs on the subject property both before and after installation of the proposed sign.
- (8) Payment of the sign permit fee, as established from time to time by the Common Council.
- (9) A written statement that all temporary signs will be removed per the limits.
- (10) Any other information that may reasonably be requested by the Zoning Administrator for the purpose of application evaluation.
- (11) Any existing or proposed sign on property abutting a state highway, United States highway, or interstate highway shall comply with all required approval from the Wisconsin Department of Transportation and the Federal Highway Administration.
- D. Granting and issuance.
 - (1) The Zoning Administrator shall review the application to ensure it is complete per the requirements of Subsection B, above.
 - (2) In cases where no other review or approvals are required under this article, the Zoning Administrator shall review said application for compliance with Subsection D, below, and shall, in writing, either approve or deny said sign permit.
 - (3) When a proposed sign is associated with any development that requires a site plan, a sign permit shall not be granted prior to the approval of a site plan. In such cases, the Zoning Administrator shall review said application for compliance with Subsection D, below, and shall schedule the item on the appropriate meeting agenda(s) or action by the body with recommending or approval authority, the Zoning Administrator shall approve or deny said sign permit based on such recommendation or action.

paid.

- (5) A granted sign permit shall expire, and shall be null and void, if the sign is not attached or erected within 180 days after the issuance of the sign permit.
- E. Basis for granting a sign permit. In deciding whether or not to grant a sign permit, the Zoning Administrator shall determine whether the proposed sign is in compliance with the provisions of this article. In such review, the Zoning Administrator may also consider the following factors:
 - (1) Whether the sign is designed, constructed, installed, or maintained in such a manner that it does not endanger public safety or traffic safety.
 - (2) Whether the sign is in compliance with all provisions of the City of Monona Municipal Code and Building Code, including those related to traffic safety, traffic visibility, sign setbacks, and structural integrity.
- F. Enforcement and revocation of sign permit.
 - (1) A sign permit may be revoked if the applicant has failed to comply with the provisions of this article or any conditions that may have accompanied the permit at the time of issuance. Revocation requires written notice by either the Zoning Administrator for Zoning Ordinance violations or the Building Inspector for Building Code or other construction code violations.
 - (2) In the event that construction, installation, or manufacture of a sign for which a permit has been issued has not commenced within 180 days from the date of the issuance of such permit, said permit shall be null and void and automatically revoked. If work authorized by such permit is suspended or abandoned for a period of 90 days any time after the work is commenced, the original permit shall become null and void. In such cases, a new permit shall be obtained to complete the work and a new permit fee shall be required.
 - (3) Any sign subject to a revoked permit shall be removed by the licensee, sign owner, or property owner within 45 days of such revocation.
 - (4) Revocation shall not result in total or partial reimbursement of permit fees paid.
- G. Appeals. Any person affected by a decision of the Zoning Administrator may petition for a hearing before the Zoning Board of Appeals. The filing of such petition automatically stays removal of any sign involved and already legally erected until the Zoning Board of Appeals decides whether to sustain, modify, or withdraw the notice.
- H. Removal of signs in violation of this article.
 - (1) If the Zoning Administrator determines that any sign exists in violation of this article, the Zoning Administrator shall notify the sign permit holder or the owner of the property on which the sign is located. Said notification shall indicate that such violation shall be corrected within 60 days of receipt of said notice on penalty of automatic revocation of

any sign permit, and that removal of the sign by the City may occur at the expense of the owner of the property.

- (2) If notification is sent and the violation is not corrected within 60 days, the Zoning Administrator shall revoke the permit for any sign which is in violation of this article. It shall be the duty of the Zoning Administrator to cause removal of such sign.
- (3) The expense of removing such sign shall be charged to the owner of the property on which the sign is located. If the owner fails to pay such expense within one month of being billed therefore, or has not made arrangement for payment satisfactory to the City Attorney, then such expense shall be charged against the property as a special charge pursuant to § 66.0627 of the Wisconsin Statutes, as may be amended from time to time.
- (4) Any sign illegally placed in a public right-of-way shall be subject to immediate removal and confiscation without notice by the Zoning Administrator.

§ 480-87. through § 480-89. (Reserved)

§ 480-90. Sign prohibitions and limitations.

The regulations contained in this section apply to signs in all zoning districts.

- A. Sign prohibitions.
 - (1) No sign shall be erected at any location where it may, by reason of its position, shape, color or design, interfere with, obstruct the view of, or be confused with any authorized traffic sign, signal or device, nor shall such sign make use of words such as "stop," "look," "danger," or any other word, phrase, symbol, or character in such a manner as to interfere with, mislead, or confuse users of streets or highways.
 - (2) No fluttering, undulating, swinging, rotating, or otherwise moving signs such as windsocks, motorized signs, pennants, and streamers shall be permitted. This shall not apply to banner signs as defined in § 480-64B(2).
 - (3) No signs shall project above the building parapet or eave, or be mounted on or extend above the roof, if attached to the building.
 - (4) No flashing, scrolling, or animated signs shall be permitted. Flashing is defined as changing more than once within a ten-second time interval.
 - (5) No signs shall be mounted to an unlicensed trailer or other unlicensed vehicle.
 - (6) No beacons or search beacons shall be permitted.
 - (7) No billboards or off-premise advertising signs shall be permitted.
 - (8) No abandoned signs shall be permitted. See § 480-64E(1).
 - (9) No illuminated sign shall be permitted unless the illumination of the sign is so designed that the lighting element is not visible from any property within a single-family, two-

family, or multifamily Zoning District.

- B. Sign limitations.
 - (1) No sign shall be erected, relocated, or maintained so as to prevent free ingress to or egress from any door, window, or fire escape, and no sign shall be attached to a standpipe or fire escape.
 - (2) No sign shall be placed in a manner that would impede vehicular or pedestrian safety, or impede access or visibility. Signs shall meet the requirements of § 480-9.
 - (3) No private sign shall be attached to or painted on any natural feature (e.g., tree or rock), fence, fire hydrant, public utility pole, public light pole, or traffic regulatory structure, unless otherwise authorized by the Director of Public Works.
 - (4) Signs must be placed on-premises which includes location on-site or within the rightof-way adjacent to the site's street frontage (outside of the vision triangle). They must be in a location approved by staff through a revocable occupancy permit granted by the City.
 - (5) No person shall: paste, tape, staple, or otherwise fasten any paper or other material to, nor paint, stencil, or otherwise write or color any object, vegetation, or pavement located within any street right-of-way; nor shall any of such object, vegetation, or pavement be defaced in any manner. The only exception to these restrictions is that painting may be allowed on curbs when approved by the Common Council.

§ 480-91. (Reserved)

§ 480-92. Appearance, construction and maintenance of signage.

- A. All signs shall be constructed, mounted, and maintained so as to comply with the appropriate provisions of the Building Code as adopted by the City relating to the design, structural members and connections. Signs shall also comply with the applicable provisions of the Electrical Code as adopted by the City.
- B. The base or support(s) of all pylon or marquee signs shall be securely anchored to a concrete base or footing. All signs shall meet minimum wind load capabilities of 30 pounds per square foot when covered with 1/2 inch of ice.
- C. The footing and related supporting structure of a permanent freestanding sign or institutional information sign, including bolts, flanges, and brackets, shall be concealed by landscaping as approved by the Plan Commission. Said landscaping shall be diverse and colorful, and shall be continuously maintained to comply with the approved landscaping plan.
- D. All pylon signs 20 feet in height or taller shall be required to submit sign construction plans stamped by a licensed engineer for approval by the Building Inspector.
- E. Signs shall be mounted so that the method of installation is concealed. Signs applied to masonry surfaces shall be mechanically fastened to mortar joints only and not directly into

brick or stone. Drilling to provide electrical service shall follow the same rule.

- F. No sign shall be suspended in a way that will allow the sign to swing due to wind action, except for arm and post signs. Signs shall be anchored to minimize any lateral movement that would cause wear on the sign face or supporting members or connections.
- G. All permanent signs and their supporting members shall be constructed of standardized, durable, all-weather sign materials.
- H. Sign materials shall be compatible with the design of the face of the facade where they are placed and shall contribute to the legibility of the sign.
- I. No combustible materials other than approved plastics shall be used in the construction of electric signs.
- J. All signage within the jurisdiction of this article shall remain in a state of proper maintenance. Proper maintenance shall be the absence of loose materials (including peeling paint, paper or other material), the lack of excessive rust, the lack of excessive vibration or shaking, and the presence of the original structural integrity of the sign, its frame and other supports, its mounting, and all components thereof.
- K. Every sign requiring either Plan Commission or Zoning Administrator approval hereafter erected shall have marked in a conspicuous place thereon the date of erection, the manufacturer's name, the permit number, and the voltage of any electrical apparatus used in connection therewith.
- L. The repainting, changing of parts, and preventive maintenance of signs which completely conform to the requirements of this article, and result in absolutely no change in the appearance of the sign from that originally approved, shall not be deemed alterations requiring a sign permit.
- M. Any signs which may be, or may hereafter become, rotted, unsafe, or in a state which is not properly maintained shall be subject to the provisions of § 1-4.

DECLARATION OF CONDOMINIUM FOR BESA CONDOMINIUM

FINAL DRAFT FOR REVIEW

Return to: Nathan J. Wautier Reinhart Boerner Van Deuren s.c. P.O. Box 2018 Madison, WI 53701-2018

_1 [__

Parcel Number

This is not a conveyance under Section 77.21(1), Wis. Stats. and is not subject to the Wisconsin real estate transfer fee or return.

There are no objections to this condominium with respect to Chapter 703 Wis. Stats. and it is hereby approved for recording.

Director of Planning and Community	& Economic Development
City of Monona	

Date

TABLE OF CONTENTS
ARTICLE I
DECLARATION
1
ARTICLE II
NAME; DESCRIPTION OF PROPERTY
1
ARTICLE III
DESCRIPTION OF UNITS
1
ARTICLE IV
COMMON ELEMENTS; LIMITED COMMON ELEMENTS
3
ARTICLE V
PERCENTAGE INTERESTS; VOTING
6
ARTICLE VI
CONDOMINIUM ASSOCIATION
7
,
ARTICLE VII
ALTERATIONS AND USE RESTRICTIONS
13
ARTICLE VIII
INSURANCE
19
ARTICLE IX
RECONSTRUCTION, REPAIR OR SALE IN
THE EVENT OF DAMAGE OR DESTRUCTION

0	0
2	2

ARTICLE X

CONDEMNATION
25

ARTICLE XI

MORTGAGEES
26

ARTICLE XII

AMENDMENT	•••
28	

ARTICLE XIII

REMEDIES
28

ARTICLE XIV

GENERAL
29

DECLARATION OF CONDOMINIUM

THIS DECLARATION OF CONDOMINIUM (this "Declaration"), is made this [___] day of [_____], 2024, by BESA MONONA, LLC, a Wisconsin limited liability company (the "Declarant").

ARTICLE I DECLARATION

Declarant hereby declares that it is the sole owner of the Land (as defined in Section 2.02), together with all improvements to be constructed thereon and all easements, rights and appurtenances pertaining thereto (the "Property"), and further declares that the Property is hereby submitted to the condominium form of ownership as provided in Chapter 703, Wisconsin Statutes (the "Condominium Ownership Act").

ARTICLE II NAME; DESCRIPTION OF PROPERTY

1. <u>Name</u>. The name of the condominium created by this Declaration (the "Condominium") is "Besa Condominium."

2. <u>Legal Description</u>. The land comprising the Property (the "Land") is located in the City of Monona, County of Dane, State of Wisconsin, and is legally described on Exhibit A attached hereto and made a part hereof.

3. <u>Address</u>. The address of the Condominium is 6501 Bridge Road, Monona, Wisconsin 53713.

ARTICLE III DESCRIPTION OF UNITS

1. <u>Identification of Units</u>. The Condominium shall consist of Eighteen (18) units (individually a "Unit" and collectively the "Units") located in the building (the "Building") identified on the condominium plat attached hereto as Exhibit B and made a part hereof (the "Condominium Plat"), together with the Common Elements as described in Article IV. The Condominium Plat shows floor plans for each Unit showing the layout, boundaries and dimensions of each Unit. The Units shall be identified as

XXX, inclusive; as numbered on the Condominium Plat. Each owner of a Unit is referred to as a "Unit Owner." Where a Unit has been sold under a land contract, the purchaser (and not the vendor) shall be the Unit Owner.

2. <u>Boundaries of Units</u>. The boundaries of each Unit shall be as follows:

(a) <u>Upper Boundary</u>. The upper boundary of the Unit shall be the interior lower surface of the supporting members of the ceiling above the highest level of the living area, extended to an intersection with the perimetrical boundaries.

(b) <u>Lower Boundary</u>. The lower boundary of the Unit shall be the upper surface of the unfinished floor of the lowest level of the Unit extended to an intersection with the perimetrical boundaries.

(c) <u>Perimetrical Boundary</u>. The perimetrical boundaries of the Unit shall be vertical planes of the inside surface of the studs supporting the interior walls, in either case extending to intersections with each other and with the upper and lower boundaries.

3. <u>Description of Units</u>. It is intended that the surface of each plane described above (be it drywall, tiles, wallpaper, paneling, carpeting or otherwise covered) is included as part of each defined Unit. The Unit shall include, without limitation, all improvements now of hereafter located within such boundaries, including:

(a) Windows and doors (with all opening, closing and locking mechanisms and all hardware) which provide direct access to or within the Unit.

(b) Interior lights and light fixtures.

(c) Cabinets.

(d) Floor, wall, baseboard, or ceiling electrical outlets and switches and the junction boxes serving them.

(e) Telephone, cable television, computer, internet, stereo or other sound systems, if any, including outlets, switches, hardware and other appurtenances serving them.

(f) Plumbing fixtures, hot water heaters, fire sprinklers, if any, water softeners, if any, and the piping, valves, and other connecting and controlling mechanisms and devices lying between the fixture and water or sewage lines serving more than one (1) Unit.

(g) The heating, ventilating and air conditioning system, including the furnaces, air conditioning equipment, the control mechanisms, all vents from the Unit to the exterior of the Condominium, including vents for furnaces, clothes dryer, range hood, all other exhaust fans, and such other vents appurtenant to each Unit, condensers and all connections thereto serving each Unit.

Specifically not included as part of a Unit are those structural components of each Building and any portion of the plumbing, electrical or mechanical systems of the Building serving more than one (1) Unit, even if located within the Unit. Any structural components, plumbing, electrical, mechanical and public or private utility lines running through a Unit that serve more than one Unit are Common Elements.

4. <u>Unit Designations</u>. Unit E100 is referred to in this Declaration as "Commercial Unit."

Units A101-A105 Units B101-B104 Units C101-C104 Units D101-D103 are referred to in this Declaration as "Live/Work Units."

3rd and 4th Floors are referred to in this Declaration as "Residential Units."

ARTICLE IV

COMMON ELEMENTS; LIMITED COMMON ELEMENTS

1. <u>Common Elements</u>. The common elements (the "Common Elements") are all of the Condominium except for the Units. The Common Elements include, without limitation, the following:

(a) The Land;

(b) The paved walkways and entrance areas situated on the Land;

(c) The foundations, columns, pilasters, girders, beams, supports, main walls (which shall be defined as exterior walls and surfaces, structural walls, roof trusses and roofs;

(d) That part of the fire sprinkler system, if any, and its associated piping and operating mechanisms serving more than one Unit;

(e) Any other portion of the improvements to the Land which is not included within the boundary of a Unit as described above;

- (f) Interior corridors serving more than one Unit;
- (g) The Building elevator;
- (h) Stair "A" serving all Units
- (i) The vehicular access areas

(j) All rights and obligations under any cross-easements benefiting the Property.

2. <u>Limited Common Elements</u>. Certain Common Elements as described in this Section shall be reserved for the exclusive use of the Unit Owners of one or more but less than all of the Units. Such Common Elements shall be referred to collectively as "Limited Common Elements." The following Common Elements shall be reserved for the exclusive use of one or more Unit Owners as described herein:

(a) All rooftop patios or balconies, if any and as applicable, attached to a given Unit and as designated and reserved on the Condominium Plat for such Unit;

(b) The parking spaces identified on the Condominium Plat as designated and reserved for any Unit; and

(c) The mailbox that exclusively serves such Unit.

- (d) Interior corridors serving more than one Unit on Floors 3 and 4;
- (e) Stair "B" serving Residential Units

(f) Exterior area on grade reserved and noted for "Future Parking" is dedicated to the Commercial Unit for potential future use at the sole expense of the Unit Owner. All future maintenance of this area will be the sole responsibility of the unit owner.

3. <u>Conflict Between Unit Boundaries; Common Element Boundaries</u>.

(a) If any portion of the Common Elements shall encroach upon any Unit, or if any Unit shall encroach upon any other Unit or upon any portion of the Common Elements as a result of the duly-authorized construction, reconstruction or repair of a Building, or as a result of settling or shifting of a Building, such that all of the actual as-built boundaries of any such Units and Common Elements are located within six (6) inches of the locations of the corresponding boundaries of such Units and Common Elements as described in Section 3.02 or elsewhere in this Declaration or as shown on the Condominium Plat, then the existing physical boundaries of such Units or Common Elements shall be conclusively presumed to be the boundaries of such Units or Common Elements, regardless of the variations between the physical boundaries described in Sections 3.02 and 3.03 or elsewhere in this Declaration or shown on the Condominium Plat and the existing physical boundaries of any such Units or Common Elements.

(b) If any portion of the Common Elements shall encroach upon any Unit, or if any Unit shall encroach upon any other Unit or upon any portion of the Common Elements as a result of the duly-authorized construction, reconstruction or repair of a Building, or as a result of settling or shifting of a Building, then a valid easement for the encroachment and for its maintenance shall exist so long as such Building stands; provided, however, that if any such encroachment or easement materially impairs any Unit Owner's enjoyment of the Unit owned by such Unit Owner or of the Common Elements in the judgment of the board of directors of the Association (as defined below), such encroachment shall be removed or just compensation shall be provided to each injured Unit Owner within ninety (90) days of the discovery of the encroachment.

(c) Following any change in the location of the boundaries of the Units under this Section 4.03, the square footages of all affected Units or Common Elements shall continue to be determined by the square footages, if any, shown on the Condominium Plat for all purposes under this Declaration.

ARTICLE V PERCENTAGE INTERESTS; VOTING

1. <u>Percentage Interests</u>. The undivided percentage interest in the Common Elements (the "Percentage Interest") appurtenant to each Unit shall be as follows:

Commercial Unit shall be Sixteen (16) percent. Live/Work Units shall be Three point Six Eight (3.68) percent. Residential Unit shall be Twenty-Five (25) percent.

With the following exception: Elevator Maintenance and repair: Commercial Unit shall be Six (6) percent. Live/Work Units shall be Zero point Three Seven Five (0.375) percent. Residential Unit shall be Eighty Eight (88) percent.

2. <u>Conveyance, Lease or Encumbrance of Percentage Interest</u>. Any deed, mortgage, lease or other instrument purporting to convey, encumber or lease any Unit shall be deemed to include the Unit Owner's Percentage Interest even though such interest is not expressly described or referred to therein.

<u>Voting</u>. Each Unit shall have the following votes, appurtenant to such Unit at meetings of the Association (as defined in Article VI).
 Commercial Unit shall have Four (4) votes
 Live/Work Units shall have One (1) vote per Live/Work Unit
 Residential Unit shall have Seven (7) votes total

4. <u>Multiple Owners</u>. If there are multiple owners of any Unit, their votes shall be counted in the manner provided in the Bylaws.

5. <u>Limitations on Voting Rights</u>. No Unit Owner shall be entitled to vote on any matter submitted to a vote of the Unit Owners until the Unit Owner's name and current mailing address, and the name and address of the Mortgagee of the Unit, if any, has been furnished to the secretary of the Association. The bylaws of the Association may contain a provision prohibiting any Unit Owner from voting on any matter submitted to a vote of the Unit Owners if the Association has recorded a statement of condominium lien on the Unit and the amount necessary to release the lien has not been paid at the time of the voting.

No voting matter may be brought to the Association, Board, or Unit owners that would affect the operations of the Commercial Unit, their users, employees, parking, or other components of business unless such actions by the unit owner are in direct and documentable violation of laws or ordinances and pose a health or life safety hazard to other Owners and users of the property.

ARTICLE VI CONDOMINIUM ASSOCIATION

1. <u>General</u>. Following the conveyance of the first Unit to any person other than Declarant, all Unit Owners shall be entitled and required to be a member of an association of Unit Owners known as "Besa Unit Owners Association, Inc." (the "Association"), which shall be responsible for carrying out the purposes of this Declaration, including exclusive management and control of the Common Elements and facilities of the Condominium, which may include the appointment and delegation of duties and responsibilities hereunder to a committee or subcommittee commissioned by the Association for that purpose. The Association shall be incorporated as a nonprofit corporation under the laws of the State of Wisconsin. The powers and duties of the Association shall include those set forth in the Association's articles of incorporation (the "Articles") and bylaws (the "Bylaws"), the Condominium Ownership Act, this Declaration and Chapter 181, Wisconsin Statutes (the "Wisconsin Nonstock Corporation Law"). All Unit Owners, tenants of Units and all other persons and entities that in any manner use the Property or any part thereof shall abide by and be subject to all of the provisions of all rules and regulations of the Association (collectively, the "Rules and Regulations"), this Declaration, the Articles and Bylaws. The Association shall have the exclusive right to promulgate, and to delegate the right to promulgate, the Rules and Regulations from time to time and shall distribute to each Unit Owner, the updated version of such Rules and Regulations upon any amendment or modification to the Rules and Regulations. Any new rule or regulation or any revision to an existing rule and regulation shall become effective immediately upon distribution to the Unit Owners.

2. <u>Declarant Control</u>. Notwithstanding anything contained in this Declaration to the contrary, the Declarant shall totally govern the affairs of the Condominium and pay all expenses thereof until a Unit has been conveyed to any person other than the Declarant. The Declarant may exercise any rights granted to, or perform any obligations imposed upon, Declarant under this Declaration through its duly authorized agent. After a Unit has been conveyed to any person other than the Declarant, except as provided in Section 6.03, the Declarant shall have the right to appoint and remove the officers of the Association and to exercise any and all of the powers and responsibilities assigned to the Association and its officers by the Articles, Bylaws, the Condominium Ownership Act, this Declaration and the Wisconsin Nonstock Corporation Law from the date the first Unit of this Condominium is conveyed by the Declarant to any person other than Declarant, until the earliest of: (a) three (3) years from such date; or (b) thirty (30) days after the conveyance of seventy five percent (75%) of the Percentage Interests; or (c) thirty (30) days after the Declarant's election to waive its right of control.

3. <u>Board of Directors</u>. The affairs of the Association shall be governed by a board of directors (the "Board of Directors"). Within thirty (30) days after the conveyance of twenty five percent (25%) of the Percentage Interests to purchasers, the Association shall hold a meeting, and the Unit Owners other than the Declarant shall elect at least twenty five percent (25%) of the directors on the Board of Directors. Within thirty (30) days after the conveyance of fifty percent (50%) of the Percentage Interests to purchasers, the Association shall hold a meeting, and the Unit Owners other than the Declarant shall elect at least thirty three and one third percent (33 1/3%) of the directors on the Board of Directors.
4. <u>Maintenance and Repairs</u>.

(a) <u>Common Elements</u>. The Association shall be responsible for the management and control of the Common Elements and Limited Common Elements (except balconies) and shall maintain the same in good, clean and attractive order and repair. In addition, the Association shall be responsible for maintaining all Limited Common Elements (except balconies and rooftop patios), for snow plowing all sidewalks, driveways, private streets or alleys, walkways, parking areas, the maintenance, repair and replacement of all outdoor amenities, including lawns, landscaping, sidewalks, bicycle paths, driveways and common exterior entrances. The Association shall be responsible for replacing when necessary any Common Elements and Limited Common Elements.

(b) Units. Each Unit Owner shall be responsible for the maintenance, repair, and replacement of all other improvements constructed within the Unit (including the electrical, heating and air conditioning systems serving such Unit, and including any ducts, vents, wires, cables or conduits designed or used in connection with such electrical, heating or air conditioning systems), and for the maintenance (but not the repair or replacement) of balconies or rooftop patios, if any, appurtenant to the Unit, except to the extent any repair cost for any of the foregoing is paid by the Association's insurance policy described in Section 8.01. Each Unit shall at all times be kept in good condition and repair. If any Unit or portion of a Unit for which a Unit Owner is responsible falls into disrepair so as to create a dangerous, unsafe, unsightly or unattractive condition, or a condition that results in damage to the Common Elements, the Association, upon fifteen (15) days' prior written notice to the Unit Owners of such Unit, shall have the right to correct such condition or to restore the Unit to its condition existing prior to the disrepair, or the damage or destruction if such was the cause of the disrepair, and to enter into such Unit for the purpose of doing so, and the Unit Owners of such Unit shall promptly reimburse the Association for the cost thereof. All amounts due for such work shall be paid within ten (10) days after receipt of written demand therefor, or the amounts may, at the option of the Association, be levied against the Unit as a Special Assessment under Section 6.07.

(c) <u>Damage Caused by Unit Owners</u>. To the extent (i) any cleaning, maintenance, repair or replacement of all or any part of any

Common Elements or the Unit is required as a result of the negligent, reckless or intentional act or omission of any Unit Owner, tenant or occupant of a Unit, or (ii) any cleaning, maintenance, repair, replacement or restoration of all or any part of any Common Element or the Unit is required as a result of an alteration to a Unit by any Unit Owner, tenant or occupant of a Unit, or the removal of any such alteration (regardless of whether the alteration was approved by the Association or any committee thereof) or (iii) the Association is required to restore the Common Elements or the Unit following any alteration of a Common Element or Limited Common Element required by this Declaration, or the removal of any such alteration, the Unit Owner that committed the act or omission or that caused the alteration, or the Unit Owners of the Unit occupied by such tenant or occupant or responsible for such guest, contractor, agent or invitee, shall pay the cost of such cleaning, maintenance, repair, replacement and restoration.

5. <u>Common Expenses</u>. Any and all expenses incurred by the Association in connection with the management of the condominium, maintenance of the Common Elements and other areas described in Section 6.04 and administration of the Association shall be deemed to be common expenses (the "Common Expenses"), including, without limitation, expenses incurred for: landscaping and lawn care; snow shoveling and plowing; improvements to the Common Elements; common grounds security lighting; municipal utility services provided to the Common Elements; trash collection; and maintenance and management salaries and wages.

6. <u>General Assessments</u>. The Association shall levy monthly general assessments (the "General Assessments") against the Unit Owners for the purpose of maintaining a fund from which Common Expenses (other than those that are the subject of Special Assessments) may be paid. The General Assessments against the Unit Owners shall be assessed in proportion to their Percentage Interests, except that until occupancy permits have been issued for all Units, the General Assessments for insurance premiums shall be levied evenly against all Units for which occupancy permits have been issued. General Assessments shall be due in advance on the first day of each month, or in such other manner as the Association may set forth in the Bylaws. Any General Assessment not paid when due shall bear interest until paid, as set forth in the Bylaws and, together with interest, collection costs, and reasonable attorneys' fees, shall constitute a

lien on the Unit on which it is assessed if a statement of condominium lien is filed within two (2) years after the assessment becomes due as provided in the Condominium Ownership Act. Notwithstanding the foregoing, Units owned by Declarant for which occupancy permits have not been issued shall not be subject to General Assessments. If, however, during the period of Declarant control the General Assessments against any Unit not owned by Declarant would exceed the amount set forth in the budget per Unit (excluding any portion of General Assessments to fund reserves), Declarant shall either: (a) record a document to cause its Units to be subject to General Assessments; or (b) pay to the Association the amount necessary to cause the General Assessments against the Unit not owned by Declarant to be reduced to the amount set forth in the budget per Unit (excluding any portion of General Assessments used to fund reserves).

7. <u>Special Assessments</u>.

(a) General. The Association may, whenever necessary or appropriate, levy special assessments (the "Special Assessment(s)") against the Unit Owners, or any of them, for deficiencies in the case of destruction or condemnation as set forth in Section 9.05 and Section 10.05; for defraying the cost of improvements to the Common Elements; for the collection of monies owed to the Association under any provision of this Declaration, including, without limitation, Section 6.04 and Article XIII, or for any other purpose for which the Association may determine a Special Assessment is necessary or appropriate for the improvement or benefit of the Condominium.

(b) Improvements that Benefit Only One Class of Units. Following completion of initial construction of the Condominium, the Association may, with the written consent or signed petition of those Residential Unit Owners or Commercial Unit Owners holding at least fiftyone percent (51%) of the votes appurtenant to the Residential Units or the Commercial Units, respectively, levy Special Assessments against the Residential Units or the Commercial Units for the purpose of:

i.making improvements that benefit only the Residential Unit Owners as a class or only the Commercial Unit Owners as a class, and for the purpose of

ii.maintaining a fund to insure, maintain or manage any improvements made under (i), above.

In either case described in (i) or (ii), the Special Assessments shall be levied equally against all of the Units within such class and shall not be levied against any of the Units in the class not benefited without the consent of the Unit Owners in such class who hold 51% of the votes appurtenant to such class.

EXAMPLE: If Residential Unit Owners holding 51% of the votes appurtenant to the Residential Units consent to upgraded wall art in the common hallways and lobbies serving the Residential Units exclusively, then the Association may levy Special Assessments against all Residential Unit Owners equally to maintain a fund to pay for insuring the upgraded wall art. The Commercial Units shall not be subject to such Special Assessments without the consent of those Commercial Unit Owners holding 51% of the votes appurtenant to the Commercial Units.

(c) Collection. Special Assessments shall be paid at such time and in such manner as the Association may determine. Any Special Assessment or installment not paid when due shall bear interest until paid, as set forth in the Bylaws and, together with the interest, collection costs and reasonable attorneys' fees, shall constitute a lien on the Unit on which it is assessed if a statement of condominium lien is filed within two (2) years after the Special Assessment becomes due as provided in the Condominium Ownership Act.

8. <u>Common Surpluses</u>. In the event that the surpluses of the Association (the "Common Surpluses") should be accumulated, other than surpluses in any construction fund as described in Section 9.06 and Section 10.06, such Common Surpluses may be credited against the Unit Owners' General Assessments in proportion to their respective Percentage Interests or may be used for any other purpose as the Association may determine.

9. <u>Certificate of Status</u>. The Association shall, upon the written request of an owner, purchaser or Mortgagee of a Unit, issue a certificate of status of lien. Any such party may conclusively rely on the information set forth in such certificate.

10. <u>Management Services</u>. The Association shall have the right to enter into a management contract with a manager selected by the Association (the "Manager") under which services may be provided to the Unit Owners. The management contract shall be subject to termination by the Association under Section 703.35 of the Wisconsin Statutes.

11. <u>Utilities and Real Estate Taxes</u>. Each Unit Owner shall be responsible for and must pay real estate and personal property taxes and special assessments assessed or levied against the Unit. Each Unit Owner shall also be responsible for the costs of all utilities used or consumed within each Unit, including, without limitation, all electric, gas, water, sewer service, and all telephone, cable, data and similar services provided to each Unit, which shall be separately metered and billed directly by the provider to the Unit Owner.

ARTICLE VII ALTERATIONS AND USE RESTRICTIONS

1. <u>Unit Alterations</u>.

(a) A Unit Owner may make improvements and alterations within its Unit; provided, however, that such improvements or alterations shall not (1) impair the structural soundness or integrity or lessen the structural support of any portion of the Condominium, (2) does not impair any easement, or (3) affect in any way the horizontal and vertical sound control measures in place as constructed. A Unit Owner may not change the dimensions of or the exterior appearance of a Unit or any portion of the Common Elements without obtaining the prior written permission of the Association, which permission may be denied in the sole discretion of the Association. Any approved improvement or alteration which changes the exterior dimensions of a Unit must be evidenced by recording a modification to this Declaration and the Condominium Plat before it shall be effective and must comply with the then applicable legal requirements for such amendment or addendum. Furthermore, any approved improvements or alterations must be accomplished in accordance with applicable laws and regulations, must not unreasonably interfere with the use and enjoyment of the other Units and the Common Elements, and must not be in violation of any underlying mortgage, land contract or similar security interest.

(b) A Unit Owner acquiring an adjoining part of another Unit may remove all or any part of the intervening partition wall or create doorways or other apertures therein. This may be done even if the partition wall may, in whole or in part, be a Common Element, provided that those acts do not impair the structural integrity or lessen the support of any portion of the Condominium, do not reduce the value of the Condominium, and do not impair any easement. The creation of doorways or other apertures is not deemed an alteration of boundaries.

2. <u>Relocation of Boundaries</u>.

(a) If the Unit Owners of adjoining Units desire to relocate their mutual boundary, the affected Unit Owners shall prepare and execute appropriate instruments.

(b) An amendment to the Declaration and an addendum to the Condominium Plat shall identify the Units and shall state that the boundaries between those Units are being relocated by agreement of the Unit Owners thereof. The amendment shall contain words of conveyance between those Unit Owners, and when recorded shall also be indexed in the name of the grantor and grantee, if applicable. If not stated, the prior allocation shall govern, until such time as the Unit Owners shall record an amendment to that effect with the Dane County Register of Deeds.

(c) Plats and plans showing the altered boundaries and the dimensions thereof between adjoining Units, and their identifying numbers or letters, shall be prepared. The plats and plans shall be certified as to their accuracy in compliance with Subsection 703.13(6) of the Wisconsin Statutes, by civil engineer, architect, or licensed land surveyor authorized to practice his or her profession in the State of Wisconsin.

(d) No boundaries of any Units may be relocated without the written consent of the Mortgagees of the Units affected.

(e) After appropriate instruments have been prepared and executed, those instruments shall become effective when the adjoining Unit Owners and the Association have executed them and they have been recorded with the Dane County Register of Deeds. The recording thereof shall be conclusive evidence that the relocation of boundaries did not violate the Condominium documents.

3. <u>Separation of Units</u>.

(a) A Commercial Unit may be separated into two (2) or more Units upon compliance with the provisions of this Section, provided that the Association approves the separation of such Unit (which approval may be denied in the sole discretion of the Association). The Association's President, upon written application of a Unit Owner proposing the separation of a Unit (the "Separator") and after thirty (30) days' written notice to all of the Unit Owners shall promptly present the matter to the Board of Directors. If approved, the President of the Association shall promptly prepare and execute appropriate instruments under this Section. An amendment to this Declaration and an addendum to the Condominium Plat shall assign a new identifying number to each new Unit created by the separation of a Unit, shall allocate to those Units, on a reasonable basis acceptable to the Separator and the other Unit Owners, all of the Percentage Interest and right to use the applicable Limited Common Elements. The vote in the Association formerly appertaining to the separated Unit will be allocated among the resulting Units. For this purpose, a fractional vote shall be permitted. The amendment shall reflect a proportionate allocation to the new Unit(s) of the liability for Common Expenses and right to Common Surpluses formally appertaining to the separated Unit.

(b) Plats and plans showing the boundaries and dimensions separating the new Units together with their other boundaries and their new identifying numbers or letters shall be prepared. The plats and plans shall be certified as to their accuracy and compliance with Subsection 703.13(7), Wisconsin Statutes, by a civil engineer, architect, or licensed land surveyor authorized to practice his or her profession in the State of Wisconsin.

(c) No Unit may be separated without the written consent of the Mortgagees of the Unit.

(d) After appropriate instruments have been prepared and executed, they shall be delivered promptly to the Separator upon payment by it of all reasonable costs for their preparation. Those instruments are effective when the Association, the Separator and the new Unit Owners have executed them and they are recorded with the Dane County Register of Deeds. The recording of the instruments shall be conclusive evidence that the separation did not violate any restrictions or limitations specified by this Declaration and that any reallocations were reasonable.

(e) No separation or subdivision of Residential Units shall be allowed.

- 4. <u>Expenses</u>. All expenses involved in any improvements, alterations boundary changes or Unit separations approved by the Association or permitted under this Article, whether or not completed, including all expenses to the Association, shall be borne by the Unit Owner or Unit Owners involved and may be charged as a Special Assessment to the affected Units in accordance with Section 6.07.
- 5. <u>Use and Restrictions on Use of Unit</u>.

(a) <u>Commercial Units</u>. The Commercial Unit shall be used only for legal uses and in compliance with all Municipal ordinances.

(b) <u>Live/Work Units.</u> Each Live/Work Unit shall be used for Commercial and Residential use. A commercial/office area must be dedicated on the first floor of each Live/Work unit. Restrictions for uses are provided by the Authority having Juris Diction and applicable Municipal Zoning Codes, however, the following uses are prohibited:

- 1: Wholesale and retail sales and storage of flammable and combustible liquids
- 2: Cleaning establishments that utilize combustible liquid solvents having a flash point of 140°F (60°C) or higher
- 3: Liquor stores and distributors
- 4: The storage or utilization of materials for agricultural purposes on the premises.
- 5: Corrosive personal or household products in their original packaging used in retail display.
- 6: Commonly used corrosive building materials.

7: The storage of black powder, smokeless propellant and small arms primers

(c) <u>Residential Unit</u>. The Residential Unit consists of 29 rental apartments and shall be used for residential purposes and for no other purpose unless otherwise authorized by the Association prior to the commencement of such use. A Residential Unit shall be deemed to be used for "Rental residential purposes" if it is occupied by no more than one family (defined to include persons related by birth, marriage or legal adoption) plus no more than two unrelated persons. No business, whether or not for profit, including, without limitation, any day care center, animal boarding business, products distributorship, manufacturing facility, sales office or professional practice, may be conducted from any Residential Unit. The foregoing restrictions as to residence and use shall not, however, be construed in such a manner as to prohibit a Residential Unit Owner from:

i.maintaining his or her personal professional library in his or her Unit;

ii.keeping his or her personal business or professional records or accounts in his or her Unit;

iii.handling his or her personal or business records or accounts in his or her Unit; or

iv.handling his or her personal business or professional telephone calls or correspondence from his or her Unit.

Nothing in this Section 7.05(a) shall authorize the maintaining of an office within a Residential Unit at which customers or clients customarily call and the same is prohibited.

6. <u>Nuisances</u>. No nuisances shall be allowed upon the Property, nor any use or practice that is unlawful or interferes with the peaceful possession and proper use of the Condominium by the Unit Owners or that would cause an increase in the premiums for insurance required to be maintained by the Association under Section 8.01. All parts of the Condominium shall be kept in a clean and sanitary condition, and no fire or other hazard shall be allowed to exist. No Unit Owner shall permit any use of its Unit or of the Common Elements that increases the cost of insuring the Condominium.

7. <u>Lease of Units</u>. Each Commercial or Live/Work Unit or any part thereof may be rented by written lease or short term agreement, provided that:

(a) in the case of a Live/Work Unit, the lease or agreement contains a statement obligating all users to abide by this Declaration, the Articles, the Bylaws and the Rules and Regulations, providing that the lease or agreement is subject and subordinate to the same; and

(b) in the case of a Live/Work Unit, the lease or agreement provides that any default arising out of the user's failure to abide by the Declaration, the Articles, the Bylaws and the Rules and Regulations shall be enforceable by the Association as a third-party beneficiary to the lease or agreement and that the Association shall have, in addition to all rights and remedies provided under the Declaration, the Articles, the Bylaws and the Rules and Regulations, the right to evict the user and/or terminate the lease or agreement should any such violation continue for a period of ten (10) days following delivery of written notice to the tenant specifying the violation.

The Association may withhold approval of any lease upon any reasonable basis, including, but not limited to: the failure of the lease terms to comply with all provisions of this Declaration, the Articles, the Bylaws and the Rules and Regulations; the past failure of the tenant or its guests to abide by all provisions of this Declaration, the Articles, the Bylaws and the Rules and Regulations; and the past use by the tenant or its invitees or guests of any part of the Condominium in a manner offensive or objectionable to the Association or other occupants of the Condominium by reason of noise, odors, vibrations, or nuisance. During the term of any lease or agreement of all or any part of a Unit, each Unit Owner of such Unit shall remain liable for the compliance of the Unit, such Unit Owner and all tenants of the Unit with all provisions of this Declaration, the Bylaws and the Rules and Regulations of the Association, and shall be responsible for securing such compliance from the tenants of the Unit. The Association may require that a copy of each lease of all or any part of a Unit be filed with the Association. The restrictions against leasing contained in this Section 7.07 shall not apply to

leases of Commercial Units, leases of the Units by the Declarant, leases of the Units to the Association, or leases of individual apartments contained in the Residential Unit.

8. <u>Signs</u>. No sign of any kind shall be displayed to the public view on any Unit without the written consent of the Association and, if Declarant owns at least one Unit, the Declarant. The Declarant reserves the right to erect signs, gates or other entryway features surrounded with landscaping at the entrances to the Condominium and to erect appropriate signage for the sales of Units.

9. <u>Garbage and Refuse Disposal</u>. No Unit shall be used or maintained as a dumping ground for rubbish, trash, garbage or waste. All clippings, rocks or earth must be in containers.

10. <u>Storage</u>. Outdoor storage of personal property shall not be permitted. No firewood or wood pile shall be kept outside a structure unless it is neatly stacked and screened from street view.

11. Pets are permitted in accordance with the current applicable Rules and Regulations.

12. <u>Landscaping</u>. Unit Owners may not plant any decorative plants, vegetables and shrubbery outside of their Unit without the prior written consent of the Association.

ARTICLE VIII INSURANCE

1. Fire and Extended Loss Insurance. The Board of Directors shall, obtain and maintain fire, casualty, and special form insurance coverage for the Units, the Common Elements and for the Association's service equipment, supplies and personal property. Insurance coverage for the Units and Common Elements shall be reviewed and adjusted by the Board of Directors from time to time to ensure that the required coverage is at all times provided. The insurance, if any, maintained by the Association shall be written on the Condominium's Units and Common Elements in the name of the Association as insurance trustee for the individual Unit Owners in their respective Percentage Interests, and may list each Unit Owner as an

additional insured with respect to its Unit. The policy shall contain the standard mortgagee clause, which shall be endorsed to provide that any proceeds shall be paid to the Association, as insurance trustee, for the use and benefit of any Mortgagee as its interest may appear. All premiums for such insurance shall be Common Expenses. In the event of damage to or destruction of all or part of the Condominium insured hereunder, the proceeds of the insurance shall be paid to the Association, as insurance trustee, for the Unit Owners and the Mortgagees and distributed as provided in Article IX.

2. Public Liability Insurance. The Board of Directors shall obtain and maintain a comprehensive liability insurance policy insuring the Association, its officers, directors, and the Unit Owners against any liability arising out of the maintenance, repair, ownership, or use of the Common Elements. Liability coverage shall be for at least \$1,000,000 per occurrence for personal injury and/or property damage or such higher limit as may be adopted from time to time by the Association to comport with the prevailing commercial practice. The insurance coverage shall be written on the Condominium in the name of the Association as insurance trustee for the Association, its directors and officers, and for the individual Unit Owners in their respective Percentage Interests. Such insurance policy shall contain a "severability of interest" or cross-liability endorsement which shall preclude the insurer from denying the claim of a Unit Owner because of the negligent acts of the Association or other Unit Owners. All premiums for such insurance shall be Common Expenses. Each Unit Owner shall have the right to insure its own Unit for personal benefit.

3. <u>Fidelity Insurance</u>. Subsequent to the sale by Declarant of the first Unit, the Association shall require or maintain fidelity coverage against dishonest acts by any person responsible for handling the funds belonging to or administered by the Association. The Association shall be named insured and the insurance shall be in an amount of not less than fifty percent (50%) of the Association's annual operating expenses and reserves. All premiums for such insurance shall be Common Expenses.

4. <u>Directors' and Officers' Insurance</u>. The Association shall require or maintain insurance on behalf of any person who is or was a director or officer of the Association against liability asserted against or incurred by him or her in any such capacity or arising out of his or her status as such. Such coverage shall be in the minimum amount of at least \$1,000,000, or

such higher minimum amounts as are needed in the discretion of the Association to comport with the prevailing commercial practice

5. Unit Owners' Insurance. Each Unit Owner shall insure all of its own personal property (whether or not such personal property is stored within the Unit owned by such Unit Owner or any Common Element or Limited Common Element) and any insurable portion of the Unit not covered by the Association's insurance as specified herein, and shall also maintain in effect at all times a comprehensive homeowner's liability policy. Each such policy shall name the Association as an additional insured. The homeowner's liability policy shall provide for coverage in the minimum amount of at least \$500,000 per occurrence for personal injury and/or property damage or such higher minimum as is needed in the discretion of the Association to comport with the prevailing commercial practice. Nothing shall prohibit Unit Owners from maintaining insurance with limits in excess of those maintained by the Association or with additional insured risks; provided, however, that each Unit Owner's own property insurance coverage shall be excess coverage only and the insurance obtained by the Association, as required under Section 8.01, shall at all times be primary coverage. Unit Owners are encouraged to submit copies of the disclosure materials for the Condominium to their insurance carriers in order to ensure adequate property and liability coverages.

6. <u>Mutual Waiver of Subrogation</u>. Nothing in this Declaration shall be construed so as to authorize or permit any insurer of the Association or a Unit Owner to be subrogated to any right of the Association or a Unit Owner arising under this Declaration. The Association and each Unit Owner hereby release each other to the extent of any perils to be insured against by either of such parties under the terms of this Declaration or the Bylaws, whether or not such insurance has actually been secured, and to the extent of their respective insurance coverage for any loss or damage caused by any such casualty, even if such incidents shall be brought about by the fault or negligence of either party for whose acts, omissions or negligence the other party is responsible. All insurance policies to be provided under this Article by either the Association or a Unit Owner shall contain a provision that they are not invalidated by the foregoing waiver. Such waiver shall, however, cease to be effective if the existence thereof precludes either the Association or a Unit Owner from obtaining such policy.

7. <u>Standards for All Insurance Policies</u>. All insurance policies provided under this Article VIII shall be written by companies duly qualified to do business in the State of Wisconsin, with a general policyholder's rating of at least "A" and a financial rating of at least Class VII, as rated in the latest edition of Best's Key Rating Guide, unless the Board of Directors determines by unanimous vote or unanimous written consent that any policy may be issued by a company having a different rating.

ARTICLE IX RECONSTRUCTION, REPAIR OR SALE IN THE EVENT OF DAMAGE OR DESTRUCTION

1. <u>Determination to Reconstruct or Repair</u>. If all or any part of the Common Elements (including the Limited Common Elements) become damaged or are destroyed by any cause, the damaged Common Elements shall be repaired or reconstructed except as otherwise provided in this Section 9.01.

Damage Less Than Five Percent of Replacement Cost. If the (a) cost to repair or reconstruct the damaged portion of the Condominium is less than five percent (5%) of the replacement cost of all improvements constituting the Condominium, the damaged portion of the Condominium shall be repaired or reconstructed even if the cost of such repair or reconstruction exceeds the available insurance proceeds. Acceptance by a Unit Owner of a deed to a Unit shall be deemed to be consent to the authorization to the Association to repair or reconstruct, as may in the future be needed from time to time, up to such stated dollar amount. If such authorization is challenged, whether through action taken at a meeting of the Unit Owners or otherwise, the issue of whether to repair or reconstruct shall be put to a vote of all of the Unit Owners entitled to vote, and such repair or reconstruction shall be deemed approved if all votes appurtenant to any one (1) Unit are cast in favor of such repair or reconstruction.

(b) <u>Damage Equal To or Greater Than Five Percent of</u> <u>Replacement Cost; Insurance Available</u>. If the cost to repair or reconstruct the damaged portion of the Condominium is equal to or greater than five percent (5%) of the replacement cost of all improvements constituting the Condominium, and the insurance proceeds plus five percent (5%) of the replacement cost of all improvements constituting the Condominium are sufficient to complete such repair or reconstruction, the damaged portion of the Condominium shall be repaired or reconstructed even if the cost of such repair or reconstruction exceeds the available insurance proceeds. Acceptance by a Unit Owner of a deed to a Unit shall be deemed to be consent to the authorization of the Association to repair or reconstruct, as may in the future be needed from time to time, up to the amount of the available insurance proceeds plus five percent (5%) of the replacement cost of all improvements constituting the Condominium. If such authorization is challenged, whether through action taken at a meeting of the Unit Owners or otherwise, the issue of whether to repair or reconstruct shall be put to a vote of all of the Unit Owners entitled to vote, and such repair or reconstruction shall be deemed approved if all votes appurtenant to any one (1) Unit are cast in favor of such repair or reconstruction.

(c) Damage Equal to or Greater Than Five Percent of Replacement Cost; Insurance Not Available. If the cost to repair or reconstruct the damaged portion of the Condominium is equal to or greater than five percent (5%) of the replacement cost of all improvements constituting the Condominium and insurance proceeds plus five percent (5%) of the replacement cost of all improvements constituting the Condominium are insufficient to complete such repair or reconstruction, the damaged portion of the Condominium shall be repaired or reconstructed unless within thirty (30) days of the date the Association receives repair or reconstruction estimates, the Unit Owners having seventy-five percent (75%) or more of the votes consent in writing to not repair or reconstruct the damaged portion of the Condominium. Delivery of such written consent under the circumstances described in this Section 9.01(c) shall be deemed to be consent to subject the Condominium to an action for partition.

2. <u>Plans and Specifications</u>. Any reconstruction or repair shall, as far as is practicable, be made in accordance with the maps, plans, and specifications used in the original construction of the Condominium, unless (1) a majority of the first Mortgagees (one vote per mortgaged Unit) approve of the variance from such plans and specifications; and (2) the Board of Directors authorizes the variance in the case of reconstruction of or repair to the Condominium. If a variance is authorized from the maps, plans, and specifications contained in the Condominium Plat or this Declaration, an amendment shall be recorded by the Association setting forth such authorized variance. 3. <u>Responsibility for Repair</u>. In all cases after a casualty has occurred to the Condominium (except as otherwise provided in Section 9.01), the Association has the responsibility of reconstruction and repair, and immediately shall obtain reliable and detailed estimates of the cost to rebuild or repair.

4. <u>Insurance Proceeds and Construction Fund</u>. Insurance proceeds held by the Association as trustee pursuant to Section 8.01 shall be disbursed by the Association for the repair or reconstruction of the damaged portion of the Condominium and Units. Unit Owners and Mortgagees shall not be entitled to receive payment of any portion of the insurance proceeds unless there is a surplus of insurance proceeds after the damaged portion of the Condominium has been completely restored or repaired as set forth in Section 9.06.

5. <u>Assessments For Deficiencies</u>. If the proceeds of insurance are not sufficient to defray the costs of reconstruction and repair by the Association, a Special Assessment shall be made against the Unit Owners in sufficient amounts to provide funds for the payment of such costs. Such assessments on account of damage to the Condominium shall be in proportion to each Unit Owner's Percentage Interest. All assessed funds shall be held and disbursed by the Association as trustee for the Unit Owners and Mortgagees involved.

6. <u>Surplus in Construction Funds</u>. All insurance proceeds and Special Assessments held by the Association as trustee for the purpose of rebuilding or reconstructing any damage to the Units, Common Elements or any Property taken by eminent domain are referred to herein as "Construction Funds." It shall be presumed that the first monies disbursed in payment of costs of reconstruction or repair are insurance proceeds. If there is a balance in the Construction Funds after payment of all costs of reconstruction or repair, such balance shall be divided among the Unit Owners according to their respective Percentage Interests.

ARTICLE X CONDEMNATION

1. <u>Allocation of Award</u>. Any damages for a taking of all or part of the Condominium shall be awarded as follows:

(a) Every Unit Owner shall be allocated the entire award for the taking of all or part of the respective Unit or any equipment, fixtures or improvements located therein and for consequential damages to the Unit or any equipment, fixtures or improvements located therein.

(b) If no reconstruction is undertaken, any award for the taking of Common Elements shall be allocated to all Unit Owners in proportion to their respective Percentage Interests.

2. <u>Determination to Reconstruct Condominium</u>. Following the taking of any part of the Condominium, then, if the Association determines that the Condominium can be restored to a usable whole, the Condominium shall be restored or reconstructed.

3. <u>Plans and Specifications for Condominium</u>. Any reconstruction shall, as far as is practicable, be made in accordance with the maps, plans and specifications used in the original construction of the portion unless seventy five percent (75%) of the Unit Owners and a majority of the first Mortgagees (one vote per mortgaged Unit) shall authorize a variance from such plans and specifications. In the event that a variance is authorized from the maps, plans or specifications contained in the Condominium Plat or this Declaration, an amendment shall be recorded by the Association setting forth such authorized variances.

4. <u>Responsibility for Reconstruction</u>. In all cases after a taking of all or part of the Condominium, the responsibility for restoration and reconstruction shall be that of the Association and it shall immediately obtain reliable and detailed estimates of the cost to rebuild.

5. <u>Assessments for Deficiencies</u>. If the condemnation award for the taking of the Common Elements is not sufficient to defray the costs of reconstruction by the Association, Special Assessments shall be made against the Unit Owners in sufficient amounts to provide funds for the payment of such costs. Such Special Assessments shall be in proportion to

each Unit Owner's respective Percentage Interest and shall constitute a Common Expense.

6. <u>Surplus in Construction Fund</u>. It shall be presumed that the first monies disbursed in payment of costs of reconstruction or restoration shall be from the award for taking. If there is a surplus of Construction Funds after payment of all costs of construction, such balance shall be divided among all Unit Owners in proportion to their respective Percentage Interests.

7. <u>Percentage Interests Following Taking</u>. Following the taking of all or any part of any Unit, the Percentage Interest appurtenant to any Unit shall be equitably adjusted to reflect the respective relative values of the remaining Units (or portions thereof) to all of the Units, determined without regard to the value of any improvements located within the Units other than those that were part of the Unit as originally constructed. The Association shall promptly prepare and record an amendment to the Declaration reflecting the new Percentage Interests appurtenant to the Units.

8. <u>Partition and Sale Upon Consent</u>. If following damage or destruction described in Section 9.01(c), the Unit Owners having Seventy-Five Percent (75%) or more of the votes consent to subject the Condominium to an action for partition, the Association shall record with the office of the Register of Deeds for Dane County, Wisconsin, a notice setting forth such facts, and upon the recording of such notice, the Condominium shall be subject to an action for partition, in which event the net proceeds of sale together with any amounts held by the Association as Construction Funds shall be considered as one (1) fund and shall be divided among the Unit Owners according to their respective Percentage Interests.

ARTICLE XI MORTGAGEES

1. <u>Notice</u>. Any holder of a recorded mortgage or any vendor under a recorded land contract encumbering a Unit (a "Mortgagee") or any guarantor of a recorded mortgage or land contract encumbering a Unit that has so requested of the Association in writing received by the Association's agent for service of process shall be entitled to receive timely written notice of the following matters:

(a) The call of any meeting of the membership of the Association or the Board of Directors to be held for the purpose of considering any proposed amendment to this Declaration, the Articles or the Bylaws;

(b) Any default under, any failure to comply with, or any violation of, any of the provisions of this Declaration, the Articles or Bylaws or the Rules and Regulations by the Unit Owner whose Unit is subject to the mortgage or land contract;

(c) Any physical damage to the Common Elements in an amount exceeding Five Percent (5%) of their replacement value;

(d) Any condemnation or casualty loss that affects either a material portion of the Condominium or the Unit securing the mortgage or land contract;

(e) Any sixty (60)-day delinquency in the payment of any charges and assessments owed under Article IX above by the owner of any Unit securing the mortgage or land contract;

(f) A lapse, cancellation, or material modification of any insurance policy maintained by the Association or under the land contract; and

Any proposed action that requires the consent of a specified percentage of Mortgagees.

2. <u>Amendment of Provisions Affecting Mortgagees</u>. Notwithstanding the provisions of Article XII of this Declaration, neither Section 11.01 nor any Section of this Declaration requiring the approval of any Mortgagee to any action shall be amended unless all Mortgagees have given their prior written approval. The Mortgagees of at least Fifty-one Percent (51%) of the mortgaged Units must consent to an amendment that is materially adverse to the Mortgagees' interests. If a Mortgagee does not respond within sixty (60) days after receipt of proper notice of any written proposal to amend this Declaration, such amendment shall be deemed approved by that Mortgagee, provided such notice was delivered to the Mortgagee by certified or registered mail with a "return receipt" requested.

3. <u>Owners of Unmortgaged Units</u>. Except as otherwise set forth in Section 11.02 above, whenever any provision contained in this Declaration requires the consent or approval (whether by vote or in writing) of a stated number or percentage of Mortgagees to any decision, each Unit Owner of any unmortgaged Unit shall be considered a "Mortgagee" as well as a "Unit Owner" for purposes of such provision.

4. <u>Condominium Liens</u>. Any Mortgagee who obtains title to a Unit under the remedies provided in the mortgage or land contract against the Unit or through foreclosure shall not be liable for more than six (6) months of the Unit's unpaid dues and assessments accrued before the date on which the holder acquired title.

ARTICLE XII AMENDMENT

Except as otherwise provided by the Condominium Ownership Act, or as otherwise provided in this Declaration, this Declaration may be amended with the written consent of not less than the number of Unit Owners who together hold at least two-thirds (2/3) of the total voting interests held by all Unit Owners. No Unit Owner's consent shall be effective without the consent of the first mortgagee of such Unit. So long as the Declarant owns any Unit, the consent in writing of the Declarant, its successors or assigns, shall also be required. No amendment shall alter or abrogate the rights of Declarant as contained in this Declaration. No amendment shall adversely affect the use and enjoyment of any Commercial Unit without the written consent of all Unit Owners of the Commercial Units and their Mortgagees. Copies of amendments shall be certified by the president and secretary of the Association in a form suitable for recording. A copy of the amendment shall be recorded with the Register of Deeds for Dane County, and a copy of the amendment shall also be mailed or personally delivered to each Unit Owner at its address on file with the Association. Until the initial conveyance of all Units, this Declaration may be amended by the Declarant alone for purposes of clarification and correction of errors and omissions.

ARTICLE XIII REMEDIES

The Association shall have the sole right to enforce the provisions hereof or any of its orders by proceedings at law or in equity against any person or persons violating or attempting to violate any provision of the Declaration, either to restrain or cure the violation or to recover damages, or both, for a period which shall include thirty (30) days from the date of the filing with the Association of a petition by any person who shall be a Unit Owner subject to this Declaration on the date of the filing, petitioning the Association to redress the violation or attempted violation of any of the provisions of this Declaration by any other persons. Liability among multiple owners of a Unit shall be joint and several. (Nothing herein shall be deemed to limit the rights of the City of Monona or the County of Dane to enforce any zoning codes, ordinances, regulations or other requirements which may be identical or similar to the requirements of this Declaration.) Such period of thirty (30) days shall be considered to be a period for the consideration of the petition by the Association and in the event the Association denies or fails to act upon the petition to the satisfaction of the petitioner within the thirty (30) day period, thereafter petitioner shall have the right to enforce the provisions hereof (except for the collection of charges and assessments under Article VI), to the extent that he or she shall so have petitioned, by proceedings at law or in equity against any person or persons violating or attempting to violate the provisions of this Declaration, either to restrain the violation or to recover damages, or both, provided, however, that any such person shall be a Unit Owner and commence such proceedings against such other person or persons within a period of sixty (60) days from (i) the date of the Association's denial of such petition, or (ii) the passage of the aforementioned thirty (30) day period for consideration of the petition by the Association. The Association or the petitioning Unit Owner(s), as the case may be, shall have the right to recover court costs and reasonable attorneys' fees in any successful action brought against another Unit Owner to enforce, or recover damages for a violation of, this Declaration. Any damages collected by the Association shall be distributed, first, to pay for all costs of enforcement, and secondly to the owners of the Units damaged by the violation pro rata. Notwithstanding the foregoing, if any Unit Owner fails to comply with the terms and conditions of this Declaration, and such failure continues beyond any applicable cure period, the Association shall have the right to cure on behalf of the Unit Owner and such Unit Owner shall promptly reimburse the Association for the cost thereof within ten (10) days after receipt of written demand therefor. Alternatively, the Association may, at the option of the Association, levy such amounts against the Unit as a Special Assessment under Article VI.

ARTICLE XIV GENERAL

1. <u>Utility Easements</u>. The Declarant hereby reserves for the Association acting by and in the discretion of the Board of Directors, the rights to grant to the City of Monona and County of Dane or public or semipublic utility companies, easements and rights of way for the erection, construction and maintenance of all poles, wires, pipes and conduits for the transmission of electricity, gas, water, telephone and for other purposes, for sewers, stormwater drains, gas mains, water pipes and mains, and similar services and for performing any public or quasi public utility function that the Board of Directors may deem fit and proper for the improvement and benefit of the Condominium. Such easements and rights of way shall be confined, so far as possible in underground pipes or other conduits, with the necessary rights of ingress and egress and with the rights to do whatever may be necessary to carry out the purposes for which the easement is created.

2. <u>Right of Entry</u>. By acceptance of a Condominium Deed, each Unit Owner shall have granted a right of entry and access to its Unit to the Association to correct any condition originating in its Unit and threatening another Unit or the Common Elements, to install, alter or repair mechanical or electrical services or other Common Elements in its Unit or elsewhere in the Condominium, and to maintain and repair Common Elements and other areas as described in Section 7.04. Such entry shall be made with prior notice to the Unit Owners, and shall be scheduled for a time reasonably convenient to the Unit Owners, except in the case of an emergency when injury or property damage will result in delayed entry. Such entry shall be done with as little inconvenience to the Unit Owners as practical, and any damage caused thereby shall be repaired by the Association and treated as a Common Expense, except as allocable to an individual Unit or Units for cause in the discretion of the Board of Directors.

3. <u>Notices</u>. All notices and other documents required to be given by this Declaration or by the Bylaws of the Association shall be sufficient if given to one (1) registered owner of a Unit regardless of the number of owners who have an interest therein. Notices and other documents to be served upon Declarant shall be given to the agent for service of process specified in Section 14.06. All owners shall provide the secretary of the Association with an address for the mailing or service of any notice or other documents and the secretary shall be deemed to have discharged his or her duty with respect to the giving of notice by mailing it or having it delivered personally to such address as is on file with him or her.

4. <u>Severability</u>. The provisions hereof shall be deemed independent and severable, and the invalidity or partial invalidity or unenforceability of any one provision or portion thereof shall not affect the validity or unenforceability of the remaining portion of said provision or of any other provision hereof. 5. <u>Declarant Access During Construction of Improvements</u>. During any period of construction of the Building and other improvements on the Property by the Declarant, the Declarant and its contractors, and subcontractors, and their respective agents and employees, shall have access to all Common Elements as may be required in connection with said construction and shall have easements for the installation and construction of the Building, improvements, utilities, driveways, parking areas, landscaping and other repairing or servicing of all or any part of the Condominium.

6. <u>Agent for Service of Process</u>. The Declarant shall be the agent for service of process in any action against the Association or brought under the Condominium Ownership Act. Service may be made upon the Declarant by serving XXXX provided, however, that the Board of Directors may at any time by duly-adopted resolution designate a successor resident agent for service of process. The designation of such person as agent shall become effective upon the execution and filing of a statement of change of registered agent with the Department of Financial Institutions as provided in the Condominium Ownership Act and the Wisconsin Nonstock Corporation Law.

7. <u>Assignment of Declarant's Rights</u>. The rights granted to the party named as "Declarant" in this Declaration may be assigned by a written, recorded instrument to any other party who assumes such rights, and, upon the recording of any such instrument, such assignee shall become, and succeed to all rights and powers granted to, "Declarant" under this Declaration.

8. <u>Conflicts</u>. In the event a conflict exists among any provisions of this Declaration, the Articles, the Bylaws and the Rules and Regulations, the Declaration shall prevail over the Articles, Bylaws and Rules and Regulations; the Articles shall prevail over the Bylaws and the Rules and Regulations; and the Bylaws shall prevail over the Rules and Regulations.

9. <u>Disclosure Regarding Construction</u>. Declarant discloses that until and after Declarant has sold all of the Units, there will be construction activity in connection with the construction of Units which may lead to noise, construction traffic, dust and other conditions incident to construction. Every Unit Owner, including Declarant, shall use reasonable efforts to avoid and to cause its contractors to limit disruption to other Unit Owners during all periods of construction activity relating to the Unit owned by such Unit Owner.

10. Disclosure Regarding Warranties. The Declarant shall assign to the Association upon substantial completion of each phase of construction all warranties held by the Declarant and covering any construction of the Common Elements. No warranties or representations, express or implied, including, but not limited to, the implied warranty of fitness for a particular purpose and merchantability, are made by the Declarant to any Unit Owner or other person or entity regarding the past or future performance or quality of the Common Elements, including the Limited Common Elements. Any implied warranty of workmanlike performance and that the Building or other Common Elements, including the Limited Common Elements, are or will be reasonably adequate for use and occupancy, created by Section 706.10(7), Wisconsin Statutes, which statutory section creates the above-stated implied warranties, for the conveyance of a newly constructed home or condominium, is hereby expressly disclaimed and excluded. Any other implied warranties created by common law, including, without limitation, the Declarant's duty to perform all work in a good and sufficient workmanlike manner, are also disclaimed and excluded. Any claims by the Association against a contractor to recover damages resulting from construction defects in any of the Common Elements or Limited Common Elements shall be subject to the provisions of Section 895.07, Wis. Stats.

[Execution Page Follows]

IN WITNESS WHEREOF, the undersigned have executed this Declaration of Condominium as of the day and year set forth above.

Besa Monona, LLC ("Declarant")

By:	
Jason Iverson	
Its:	

STATE OF WISCONSIN)) ss. COUNTY OF DANE)

Personally came before me this _____ day of ______ 2024, Jason Iverson, the Managing Member of Besa Monona, LLC, a Wisconsin limited liability company, who acknowledged the foregoing document for the purposes recited therein on behalf of said limited liability company.

> Name:_____ Notary Public, State of Wisconsin My Commission:_____

This document drafted by and should be returned to:

Nathan J. Wautier Reinhart Boerner Van Deuren s.c. P.O. Box 2018 Madison, Wisconsin 53701-2018 (608) 229-2200

CONSENT OF MORTGAGEE

The undersigned, being the holder of a mortgage against the Property, does hereby consent to all of the terms and conditions of the foregoing Declaration, and agrees that its interest in the Property shall be subject in all respects to the terms thereof.

Dated this ____ day of _____, 2024

	By:
	Name:
	Title:
STATE OF WISCONSIN)	
) ss.	
COUNTY OF DANE)	
Personally came before me this _	day of, 2024, the
of	, who acknowledged the
foregoing document for the purposes re-	cited therein on behalf of the same.

Name:	
Notary Public, State of Wisconsin	
My Commission:	

EXHIBIT A

LEGAL DESCRIPTION

[_____]

EXHIBIT B

CONDOMINIUM PLAT



.







.

.





•



A Condominium, A Condominium Plat. Leader In: (14). South Towne Plat. In the NW of the NW of of Societor 29. T7N, RTOE, In the City of Monone, Dane County, Waconde. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. Legal Description: Lot Fourteen (14). South Towne Plat, in the City of Monone, Dane County, Wisconsin. South County County, South Towne Plat, in the City of Monone, Dane South County County, South Towne Plat, in the City of Monone, Dane County, Wisconsin. South County County, South Towne Plat, in the City of Monone, Dane County, Wisconsin. South County County, South Towne Plat, in the City of Monone, Dane County, Wisconsin. South County County, South Towne, County, South Towne, County, South Towne, County, South Towne, South Towne	Condominium & Condominium Plat	<u> </u>	
Wisconsin. Legal Description: Lot Fourteen (14), South Towne Plat, in the City of Monona, Dane County, Wisconsin. Linted Common Element (L.C.E.) Parcet Boundary Ine Building Permiera is tack liked Street Right-drives and Plated Likes Street Right-drives and Plated Likes Street Controllines Listen Likes Listen Likes Street Controllines Listen Likes	Located In: South Towne Plat, in the NW $\frac{1}{4}$ of the NW $\frac{1}{4}$ of Section 29, T7N, R10E, in the City of Monona, Dane County,		
	Wisconsin. Legal Description: Lot Fourteen (14), South Towne Plat, in the City of Monona, Dane County, Wisconsin. Limited Common Element (L.C.E.) Parcel Boundary line Building Perimeter at each level		
			 _

•

BYLAWS

OF

BESA UNIT OWNERS ASSOCIATION, INC.
TABLE OF CONTENTS

		Page
ARTICLE	E I NAME AND ADDRESS	1
1.01	Name; Purpose	1
1.02	Address	1
1.03	Binding Effect	1
1.04	Capitalized Terms	1
ARTICLE	E II MEMBERSHIP	1
2.01	Membership	1
2.02	Commencement and Termination	1
2.03	Withdrawal or Expulsion	2
2.04	Membership Certificates	2
2.05	Membership List	2
2.06	Transfer of Membership	2
2.07	Effect of Condominium Lien	2
2.08	Quorum	3
2.09	Vote Required to Transact Business	3
2.10	Proxies	3
2.11	Voting Designations of Multiple Unit Owners	3
ARTICLE	E III MEETINGS OF MEMBERS	3
3.01	Place	3
3.02	Annual Meetings	3
3.03	Special Meetings	4
3.04	Notice of Meetings	4
3.05	Adjourned Meetings	4
3.06	Duties of Officers at Meetings	4
3.07	Order of Business	4
3.08	Action Without a Meeting by Written Consent	5
3.09	Action Without a Meeting by Written Ballot	5
ARTICLE	E IV BOARD OF DIRECTORS	5
4.01	Number and Membership in Association	5
4.02	Term of Office	6
4.03	Election of Directors	6
4.04	Vacancy and Replacement	7
4.05	Removal	7
4.06	Compensation	7
ARTICLE	E V MEETINGS OF THE BOARD OF DIRECTORS	7
5.01	Regular Meetings	7
5.02	Special Meetings	8

5.03	Notice of Special Meetings	8
5.04	Quorum	8
		Page
5.05	Order of Business	8
5.06	Action Without a Meeting by Written Consent	8
ARTICLE	VI POWERS AND DUTIES OF BOARD OF DIRECTORS	9
6.01	Powers and Duties	9
6.02	Manager	11
ARTICLE	VII OFFICERS AND THEIR DUTIES	11
7.01	Officers	11
7.02	Election of Officers	11
7.03	Term	11
7.04	Special Appointments	11
7.05	Resignation and Removal	11
7.06	Vacancies	12
7.07	Duties	12
7.08	Compensation	13
7.09	Fidelity Bonds	13
ARTICLE	VIII BOOKS AND RECORDS	13
8.01	Inspection	13
8.02	Audits	13
ARTICLE	IX BUDGET, ASSESSMENT AND ANNUAL REPORT	13
9.01	Fiscal Year	13
9.02	Budget	13
9.03	Levying and Payment of General Assessments	14
9.04	Special Assessments	14
9.05	Association Remedies Upon Nonpayment of Assessments	14
9.06	Annual Report	14
ARTICLE		15
ARTICLE	XI ENFORCEMENT OF CONDOMINIUM	
	DOCUMENTS	15
_	XII LIABILITY AND INDEMNITY	16
12.01	General Scope and Definitions	16
12.02	Mandatory Indemnification	17
12.03	Determination of Right to Indemnification	18
12.04	Allowance of Expenses as Incurred	18
12.05	Partial Indemnification	19
12.06	Indemnification of Employees and Agents	19
12.07	Limited Liability of Directors and Officers	19
12.08	Severability of Provisions	20

12.09	Nonexclusivity of Rights	20
12.10	Purchase of Insurance	20
12.11	Benefit	21
12.12	Amendment	21
		Page
ARTICLE	XIII GENERAL PROVISIONS	21
13.01	Seal	21
13.02	Interpretation	21
13.03	Notices	21
ARTICLE	22	

BYLAWS OF

BESA UNIT OWNERS ASSOCIATION, INC.

ARTICLE I

NAME AND ADDRESS

1.01 <u>Name; Purpose</u>. The name of the corporation shall be Besa Unit Owners Association, Inc. (the "Association"). The Association is incorporated as a nonstock, nonprofit corporation under the provisions of the Wisconsin Nonstock Corporation Law, Chapter 181 of the Wisconsin Statutes.

1.02 <u>Address</u>. The principal office of the Association shall be determined. This address shall also be the mailing address of the Association.

1.03 <u>Binding Effect</u>. These Bylaws (the "Bylaws") shall be binding upon the Unit Owners, their heirs, successors and assigns and shall govern the use, occupancy, operation and administration of the Condominium.

1.04 <u>Capitalized Terms</u>. Capitalized terms not defined in these Bylaws shall have the definitions given to such terms in the Declaration of Condominium for Besa Condominium, executed by Besa Monona, LLC (the "Declarant") and recorded in the office of the Dane County Register of Deeds (the "Declaration").

ARTICLE II

MEMBERSHIP

2.01 <u>Membership</u>. The membership of the Association shall at all times consist exclusively of all of the Unit Owners of the Condominium. Land contract vendees and not land contract vendors shall be members of the Association. Persons who hold an interest in a Unit merely as security for the performance of an obligation (including Mortgagees) are not members of the Association.

2.02 <u>Commencement and Termination</u>. Membership shall immediately commence upon acquisition of an ownership interest in a Unit of the Condominium and shall immediately terminate upon conveyance of such ownership interest. If a

Unit Owner's ownership interest passes to its personal representative or to a trustee upon the Unit Owner's death, such personal representative or trustee shall be a member of the Association.

2.03 <u>Withdrawal or Expulsion</u>. No Unit Owner may voluntarily withdraw or be expelled from membership in the Association.

2.04 <u>Membership Certificates</u>. Membership certificates shall not be issued.

2.05 Membership List. The Association shall maintain a current membership list listing all Unit Owners of each Unit, the current mailing address for each Unit Owner to which notice of meetings of the Association shall be sent, all Mortgagees of the Unit, if any, and, in the case of multiple owners of a Unit, the Unit Owner, if any, designated to cast any or all of the votes pertaining to such Unit in accordance with the Declaration. Each Unit Owner shall promptly provide written notice to the Association of any transfer of its Unit as provided in Section 2.06 and of any change in such Unit Owner's name or current mailing address. No Unit Owner may vote at meetings of the Association until the name and current mailing address of such Unit Owner has been provided to and received by the secretary of the Association. Any Unit Owner that mortgages its Unit or any interest therein or enters into a land contract with respect to its Unit shall notify the secretary of the name and mailing address of its Mortgagee and shall also notify the secretary when such mortgage has been released or such land contract has been fulfilled, and the secretary shall make appropriate changes to the membership list effective as of the date of the mortgage, release, land contract or fulfillment, as the case may be.

2.06 <u>Transfer of Membership</u>. Each membership shall be appurtenant to the Unit upon which it is based and shall be transferred automatically upon conveyance with the transfer of a Unit. As soon as possible following the transfer of a Unit, the new Unit Owners shall give written notice to the secretary of the Association of such transfer identifying the Unit and setting forth the names and mailing addresses of the new Unit Owners, the date of the transfer, the names and addresses of each Mortgagee, if any, and in the case of a Unit owned by multiple Unit Owners, the name of the person designated to vote, if any. The Association shall make appropriate changes to the membership list described in Section 2.05 effective as of the date of transfer.

2.07 <u>Effect of Condominium Lien</u>. No Unit Owner may vote on any matter submitted to a vote of the Unit Owners if the Association has recorded a statement of

condominium lien on the Unit owned by such Unit Owner and the amount necessary to release the lien has not been paid at the time of the voting.

2.08 <u>Quorum</u>. Unit Owners holding fiftyone percent (51%) of the total votes of the Association as set forth in the Declaration, present in person or represented by proxy, shall constitute a quorum at all meetings of the Unit Owners for the transaction of business.

2.09 <u>Vote Required to Transact Business</u>. When a quorum is present in person or represented by proxy at any meeting, a majority of the votes cast shall decide any question brought before the meeting unless the question requires a different vote by express provision in the Declaration, Articles of Incorporation of the Association (the "Articles"), Wisconsin Condominium Ownership Act, Wisconsin Nonstock Corporation Law or these Bylaws, in which case such express provision shall apply.

2.10 <u>Proxies</u>. All proxies shall be in writing, signed by the Unit Owner giving such proxy, and filed with the secretary of the Association before or at the time of the meeting. No proxy shall be valid after one hundred eighty (180) days from its date of issuance, unless granted to a Mortgagee or tenant of a Unit.

2.11 <u>Voting Designations of Multiple Unit Owners</u>. No designation of a single Unit Owner to cast any vote appertaining to any Unit owned by multiple Unit Owners shall be effective until written notice of such designation signed by all Unit Owners of such Unit has been received by the secretary of the Association prior to the casting of such vote. If any Unit Owner is so designated, then except as provided in the Declaration, only that Unit Owner shall be entitled to cast such vote in person or by proxy. A voting designation may be limited in time or may be changed by notice in writing to the secretary of the Association signed by all Unit Owners of such Unit. If no designation has been made, then any of the multiple Unit Owners present at the meeting in person or by proxy shall be entitled to cast the entire vote for the Unit, provided that if more than one multiple Unit Owner is so present, each multiple Unit Owner shall be entitled only to vote its proportionate share of the vote appurtenant to the Unit.

ARTICLE III

MEETINGS OF MEMBERS

3.01 <u>Place</u>. All meetings of the Unit Owners shall be held at a place in Dane County, Wisconsin, that shall be stated in the notice of the meeting.

3.02 <u>Annual Meetings</u>. The first annual meeting of the Unit Owners shall be held on the second Monday of the first December after the Declarant has ceased to control the Association as provided in Section 6.02 of the Declaration. Thereafter, regular annual meetings of the Unit Owners shall be held on the second Monday of December of each succeeding year.

3.03 <u>Special Meetings</u>. Special meetings of the Unit Owners may be called at any time by the president of the Association and shall be called upon the written request of Unit Owners holding at least twentyfive percent (25%) of the votes. Business transacted at special meetings shall be limited to the objects stated in the notice of such meeting.

3.04 <u>Notice of Meetings</u>. No annual or special meeting of the Unit Owners may be held except upon at least ten (10) days' (but not more than 60 days') written notice delivered or mailed to each Unit Owner at the address shown on the Association's current membership list. Such notice shall specify the place, day and hour of the meetings and, in the case of a special meeting, the purpose of the meeting. Prior notice of a meeting is not required to any Unit Owner that signs a waiver of notice of such meeting.

3.05 <u>Adjourned Meetings</u>. If a quorum shall not be present in person or represented by proxy at any meeting, the Unit Owners present shall have the power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present or represented by proxy. At such adjourned meeting at which a quorum shall be present or represented by proxy, any business may be transacted which might have been transacted at the meeting originally called.

3.06 <u>Duties of Officers at Meetings</u>. The president of the Association shall preside at all meetings of the Unit Owners, and in his or her absence, the vice president shall preside. The secretary shall take the minutes of the meeting and keep such minutes in the Association's minute book. Votes at all meetings shall be counted by the secretary.

3.07 <u>Order of Business</u>. The order of business at all meetings of the Unit Owners shall be as follows:

- (a) Calling the meeting to order.
- (b) Calling the roll of Unit Owners and certifying the proxies.
- (c) Proof of notice of meeting or waiver of notice.
- (d) Reading and disposal of any unapproved minutes.
- (e) Reports of officers.
- (f) Reports of committees (if appropriate).
- (g) Election of directors (if appropriate).
- (h) Unfinished business.
- (i) New business.
- (j) Adjournment.

3.08 <u>Action Without a Meeting by Written Consent</u>. Any action required or permitted by any provision of the Wisconsin Condominium Ownership Act, the Wisconsin Nonstock Corporation Law, the Declaration, the Articles or these Bylaws to be taken by the vote of the Unit Owners may be taken without a meeting if a written consent, setting forth the action so taken, is signed and dated by all Unit Owners that would have been entitled to vote upon the action at such meeting and that hold a number of votes equal to the greater of: (a) the number of votes that would be required to approve the matter at a meeting attended by one hundred percent (100%) of the Unit Owners entitled to vote or (b) fifty-one percent (51%) of the total number of votes in the Association.

3.09 Action Without a Meeting by Written Ballot. Any action required or permitted by any provision of the Wisconsin Condominium Ownership Act, the Wisconsin Nonstock Corporation Law, the Declaration, the Articles or these Bylaws to be taken by the vote of the Unit Owners may be taken without a meeting if the Association delivers a written ballot to every Unit Owner entitled to vote on the matter. The written ballot shall set forth each proposed action, shall provide an opportunity to vote for or against each proposed action, and shall be accompanied by a notice stating the number of responses needed to meet the quorum requirements, the percentage of approvals necessary to approve each matter other than election of directors and the time by which the ballot must be received by the secretary of the Association in order to be counted. Approval of any action by written ballot shall be valid only when the number of votes cast by ballot equals or exceeds the quorum required at a meeting authorizing the action, and the number of approvals equals or exceeds the number of votes that would be required to approve the matter at a meeting at which the total number of votes cast was the same as the number of votes cast by ballot. Once received by the secretary of the Association, a written ballot may not be revoked.

ARTICLE IV

BOARD OF DIRECTORS

4.01 Number and Membership in Association. The affairs of the Association shall be managed initially by a board of directors (the "Board of Directors") composed of three directors selected by the Declarant. At such time as the Declarant has conveyed twentyfive percent (25%) of the Percentage Interest to purchasers, the number of directors on the Board of Directors shall be increased to four (4). At such time as the Declarant has conveyed fifty percent (50%) of the Percentage Interests to purchasers, the number of directors on the Board of Directors shall be decreased to three (3). Upon the earlier of (a) the tenth anniversary of the date the Declaration is recorded, or (b) thirty days after the date Declarant has conveyed seventy-five percent (75%) of the Percentage Interests to purchasers, the number of directors on the Board of Directors shall be increased to five (5). No more than one director at any given time may be a person who is not also a Unit Owner; provided, however, that during the period of Declarant control as provided in Section 6.02 of the Declaration, any person named by the Declarant to the Board of Directors shall be deemed to be a "Unit Owner" for purposes of this requirement only and provided further, that in the case of a Unit that is owned by an entity rather than an individual, any person who is an officer, member, partner, director, employee or designee of such entity shall be deemed to be a "Unit Owner" for purposes of this requirement only.

4.02 <u>Term of Office</u>. The initial Board of Directors shall serve until the Declarant has conveyed twentyfive percent (25%) of the Percentage Interests to purchasers. Within thirty (30) days after the conveyance of twentyfive percent (25%) of the Percentage Interests to purchasers, the Unit Owners other than the Declarant shall elect one director to serve on the Board of Directors. The Declarant shall elect the remaining three directors. Such Board of Directors shall serve until the Declarant has conveyed fifty percent (50%) of the Percentage Interests to purchasers. Within thirty (30) days after the conveyance of fifty percent (50%) of the Percentage Interests to purchasers, the Unit Owners other than the Declarant shall elect to purchasers, the Unit Owners other than the Declarant shall elect to purchasers, the Unit Owners other than the Declarant shall elect to serve on the Board of Directors. The Declarant shall elect one director to serve on the Owners other than the Declarant shall elect the Percentage Interests to purchasers, the Unit Owners other than the Declarant shall elect the director to serve on the Board of Directors. The Declarant shall elect the Percentage Interests to purchasers, the Unit Owners other than the Declarant shall elect the

remaining two directors. Such Board of Directors shall serve until the next election upon expiration of the period of Declarant control as provided in Section 6.02 of the Declaration. Not later than thirty (30) days after the expiration of the period of Declarant control, a special meeting of the Unit Owners shall be called, and the Unit Owners shall elect all five (5) directors to serve on the Board of Directors. Such directors shall take office upon such election and shall serve until the first annual meeting of the Unit Owners as provided in Section 3.02. Thereafter, each director shall take office at the annual meeting and shall serve for a term of one (1) year or until his or her successor shall be elected.

4.03 <u>Election of Directors</u>. One (1) month prior to each annual meeting of the Unit Owners, the secretary of the Association shall mail to all Unit Owners a notice setting a deadline for nomination of persons to serve as directors on the Board of Directors. All nominations shall be mailed to the secretary. Unit Owners must obtain the prior consent of any person they nominate and may nominate themselves. Only Unit Owners entitled to vote upon the election of any director may nominate a person to serve as a director. If the number of nominees equals the number of directors to be elected, the nominees shall automatically become the new directors to take office at the annual meeting. If the number of nominees is fewer than the number of directors to be elected, the secretary shall solicit further nominees by mail. If the number of nominees exceeds the number of directors to be elected, the secretary shall conduct an election by written ballot in accordance with Section 3.09 with all written ballots due prior to the deadline set by the secretary. Each Unit shall have the number of votes provided in the Declaration. The persons receiving the largest number of votes shall be elected as directors and shall take office at the annual meeting.

4.04 <u>Vacancy and Replacement</u>. If the office of any director becomes vacant because of death, resignation, disqualification or removal from office, such vacancy shall be filled by vote of a majority of the remaining directors at a special meeting of the Board of Directors held for that purpose promptly after the occurrence of such vacancy, even though the directors present may constitute less than a quorum, and each person so elected shall be a member of the Board of Directors for the remainder of the term of the director who left office or until a successor is elected in accordance with these Bylaws. Notwithstanding the foregoing, during the period of Declarant control as described in Section 6.02 of the Declaration, only the Declarant shall have the right to replace any director elected by Declarant.

4.05 <u>Removal</u>. Prior to the expiration of the period of Declarant control as described in Section 6.02 of the Declaration, only the Declarant shall have the right

to remove a director from the Board of Directors. Thereafter, any director may be removed from the Board of Directors, with or without cause, by a majority vote of the Unit Owners.

4.06 <u>Compensation</u>. No director shall receive any compensation for his or her services as a director of the Association other than reimbursement for reasonable outofpocket expenses incurred in the performance of directors' duties.

ARTICLE V

MEETINGS OF THE BOARD OF DIRECTORS

5.01 <u>Regular Meetings</u>. Until the expiration of Declarant control as described in Section 6.02 of the Declaration, the regular meeting of the Board of Directors shall be held annually on the second Monday of December at the time and place designated in the notice of such meeting. Thereafter, regular meetings of the Board of Directors shall be held annually without notice following the annual meeting of the Unit Owners at the same place as the Unit Owners' meeting or at such place as the Board of Directors may vote to hold the meeting.

5.02 <u>Special Meetings</u>. Special meetings of the Board of Directors may be called at any time by the president and shall be called by the president or secretary at the request of any director on the Board of Directors. Business transacted at all special meetings shall be limited to the objects stated in the notice of such meeting.

5.03 <u>Notice of Special Meetings</u>. No special meeting of the Board of Directors may be held except upon at least three (3) days' prior written notice delivered or mailed by the secretary to each member of the Board of Directors. Such notice shall specify the place, day and hour of the meeting of the Board of Directors and the purpose of the meeting. Attendance by any director at any meeting of the Board of Directors shall be deemed a waiver of such notice.

5.04 Quorum. A majority of the Board shall constitute a quorum for the transaction of business. Except as otherwise expressly provided in the Wisconsin Condominium Ownership Act, the Wisconsin Nonstock Corporation Law, the Declaration, the Articles or these Bylaws, every act of a majority of directors present at any meeting at which there is a quorum shall be the act of the Board of Directors. If a quorum is not present at the meeting, the directors then present may adjourn the meeting until such time as a quorum is present, and at such later meeting at which a

quorum is present, may transact any business which might have been transacted at the meeting originally called.

5.05 <u>Order of Business</u>. The order of business at all meetings of the Board of Directors shall be as follows:

- (a) Calling the meeting to order.
- (b) Proof of notice of meeting or waiver of notice.
- (c) Reading and disposal of any unapproved minutes.
- (d) Reports of officers.
- (e) Reports of committees (if appropriate).
- (f) Election of officers (if appropriate).
- (g) Unfinished business.
- (h) New business.
- (i) Adjournment.

5.06 <u>Action Without a Meeting by Written Consent</u>. Any action required or permitted by the Articles or these Bylaws to be taken by the Board of Directors may be taken without a meeting if a written consent, setting forth the action so taken, is signed by two-thirds (2/3) of the directors then in office.

ARTICLE VI

POWERS AND DUTIES OF BOARD OF DIRECTORS

6.01 <u>Powers and Duties</u>. All of the powers and duties of the Association under the Declaration, the Articles, these Bylaws, the Wisconsin Condominium Ownership Act and the Wisconsin Nonstock Corporation Law shall be exercised by the Board of Directors except those powers and duties specifically given to or required of any committees of the Association or the Unit Owners. The powers and duties of the Board of Directors include, without limitation, the power or duty to:

(a) Adopt budgets for revenues, expenditures and reserves.

(b) Levy and collect General Assessments and Special Assessments and disburse funds in payment of the Association's expenses.

(c) Manage, maintain, repair, replace, improve, operate and regulate the Common Elements, Limited Common Elements, and any property owned or leased by the Association.

(d) Grant easements, licenses and rights-of-way through or over the Common Elements.

(e) Hire and supervise any property manager or agent, security manager or agent, other manager or agent, employee, attorney, accountant or any other independent contractor whose services the Board of Directors determines are necessary or appropriate.

(f) Sue on behalf of all Unit Owners.

(g) Make contracts and incur liabilities.

(h) Purchase, take, receive, rent or otherwise acquire and hold any interest in real or personal property, including any Unit of the Condominium.

(i) Sell, convey, mortgage, encumber, lease, exchange, transfer or otherwise dispose of any interest in real or personal property, including any Unit of the Condominium.

(j) Receive any income derived from payments, fees or charges for the use, rental or operation of the Common Elements and any property owned or leased by the Association.

(k) Adopt, amend and repeal rules and regulations governing the operation, maintenance and use of any portion of the Condominium and the personal conduct of any person upon or with regard to Condominium property, including the imposition of charges for the use of Common Elements and penalties for infractions of the rules and regulations of the Association. Such rules and regulations may also be adopted, amended and repealed by the Unit Owners having sixtyseven percent (67%) or more of the votes of the Association. Notwithstanding anything in these Bylaws to the contrary, (i) rules and regulations which are adopted, amended or repealed by the Unit Owners may not thereafter be amended, repealed or readopted by the Board of Directors; and (ii) the Declarant and its successors and assigns shall not be subject to or bound by any rule, regulation or amendment to a rule or

regulation that is adopted without the written consent of the Declarant and its successors and assigns to the specific rule, regulation or amendment.

(1) Insure the Condominium property and property owned or leased by the Association against loss by fire and other casualty and the Association and Unit Owners against public liability as provided in the Declaration and purchase such other insurance as the Board of Directors may deem advisable.

(m) Keep all of the books and records and prepare accurate reports of all transactions of the Association.

(n) Appoint committees to carry out any tasks which the Board of Directors deems necessary or appropriate.

(o) Designate depositories and establish accounts for the funds of the Association and determine which officers or agents shall be authorized to withdraw and transfer funds deposited in such accounts.

(p) Maintain such reserve funds for the operation, maintenance, repair and replacement of Common Elements, Limited Common Elements and any property owned or leased by the Association, for contingencies and for making up any deficit in the Common Expenses for any prior year as the Board of Directors may deem proper or as may be required by law.

(q) Delegate any or part of the powers and duties of the Board of Directors or Association officers to committees of the Association or to a manager or managing agent.

6.02 <u>Manager</u>. The Board of Directors may hire a manager or managing agent at a compensation rate established by the board to perform such duties and services as the Board of Directors shall authorize, including, without limitation, the duties enumerated in Sections 6.01 and 7.07.

ARTICLE VII

OFFICERS AND THEIR DUTIES

7.01 <u>Officers</u>. The principal officers of the Association shall be the president, vice president, secretary and treasurer, all of whom shall be elected by the

Board of Directors. All officers shall be Unit Owners, provided, however, that during the period of Declarant control as provided in Section 6.02 of the Declaration, any person named by the Declarant to the Board of Directors or as an officer shall be deemed to be a "Unit Owner" for purposes of this requirement only and provided further, that in the case of a Unit that is owned by an entity rather than an individual, any person who is an officer, member, partner, director, employee or designee of such entity shall be deemed to be a "Unit Owner" for purposes of this requirement only. The same individual may simultaneously hold more than one office in the Association.

7.02 <u>Election of Officers</u>. The first election of officers shall take place at the first meeting of the initial Board of Directors. Thereafter, the officers shall be elected annually by the Board of Directors at its regular meeting.

7.03 <u>Term</u>. Each officer of the Association shall hold office for a term of one (1) year or until his or her successor shall be elected.

7.04 <u>Special Appointments</u>. The Board of Directors may elect such other officers as the affairs of the Association may require, each of whom shall hold office for a period specified by the Board of Directors which shall not exceed three (3) years, have such authority and perform such duties as the Board of Directors may from time to time determine.

7.05 <u>Resignation and Removal</u>. Any officer may be removed from office by the Board of Directors whenever in its judgment the best interests of the Association will be served thereby. Any officer may at any time resign by giving written notice to the president or the secretary. Such resignation shall take effect on the date of receipt of such notice by the president or the secretary or at any later time specified in the notice. Unless otherwise specified in the notice, the acceptance of the resignation described in the notice shall not be necessary for its effectiveness.

7.06 <u>Vacancies</u>. A vacancy in any office may be filled by appointment by the Board of Directors. The officer appointed to fill such vacancy shall serve for the remainder of the term of the officer replaced.

7.07 <u>Duties</u>. Unless otherwise indicated by the Board of Directors or delegated to a manager or managing agent pursuant to Article VI, the duties of the officers are as follows:

(a) <u>President</u>. The president shall preside at all meetings of the members of the Association and of the Board of Directors; oversee the implementation of the Board of Directors' orders and resolutions; sign all leases, mortgages, deeds, contracts, checks, promissory notes and other written instruments on behalf of the Association; generally manage the business of the Association; supervise and direct all other officers of the Association; and perform such other duties incident to the office of president as may be required under the Wisconsin Condominium Ownership Act, the Wisconsin Nonstock Corporation Law, the Declaration, the Articles or these Bylaws, or by the Board of Directors.

(b) <u>Vice President</u>. The vice president shall act in the place of the president in the event of the president's absence, inability or refusal to act, and shall exercise and discharge such other duties as may be required by the Board of Directors.

(c) <u>Secretary</u>. The secretary shall record the votes and keep the minutes of all meetings and proceedings of the Board of Directors and of the Unit Owners; serve notices of the meetings of the Board of Directors and of the Unit Owners; keep all books and records of the Association other than books of account, including the membership list described in Section 2.05; and perform such other duties incident to the office of secretary as may be required under the Wisconsin Condominium Ownership Act, the Wisconsin Nonstock Corporation Law, the Declaration, the Articles or these Bylaws, or by the Board of Directors.

(d) <u>Treasurer</u>. The treasurer shall receive and deposit in appropriate bank accounts all monies of the Association and disburse such funds as directed by the president or by the Board of Directors; keep complete and accurate books of account; prepare the annual report of the business transacted by the Association each year; and prepare a proposed annual operating budget each year for consideration of the Board of Directors or Unit Owners.

7.08 <u>Compensation</u>. No officer shall receive any compensation for his or her services as an officer of the Association, other than reimbursement for reasonable outofpocket expenses incurred in the performance of officers' duties.

7.09 <u>Fidelity Bonds</u>. The Board of Directors may require that any officers, agents or employees of the Association handling or responsible for Association funds

shall furnish adequate fidelity bonds. The premiums of such bonds shall be paid by the Association.

ARTICLE VIII

BOOKS AND RECORDS

8.01 <u>Inspection</u>. The books, records, minutes, papers and membership list of the Association shall at all times, during reasonable business hours, be subject to inspection by any Unit Owner. The Declaration, the Articles and the Bylaws shall be available for inspection by any Unit Owner, Mortgagee or prospective purchaser of a Unit at the principal office of the Association, where copies may be purchased at reasonable cost.

8.02 <u>Audits</u>. The accounts and records of the Association shall be audited at least once every other year by an audit committee selected by the Board of Directors. The committee shall retain such professional auditors and other independent examiners as it deems appropriate. The cost of such audit shall be a Common Expense.

ARTICLE IX

BUDGET, ASSESSMENT AND ANNUAL REPORT

9.01 <u>Fiscal Year</u>. The fiscal year of the Association shall begin on the first day of January and end on the last day of December.

9.02 <u>Budget</u>. Throughout the period of Declarant control as described in Section 6.02 of the Declaration, the Board of Directors shall adopt an annual operating budget for the Association at the annual meeting of the Board of Directors, provided, however, that the first annual operating budget for the Association shall be adopted by the Board of Directors prior to the first sale of a Unit by the Declarant. After the expiration of the period of Declarant control as described in Section 6.02 of the Declaration, the Unit Owners holding at least fiftyone percent (51%) of the votes present in person or represented by proxy at their annual meeting shall adopt the annual operating budget for the Association at such annual meeting. The budget shall be effective for the period beginning January I through December 31 of the succeeding year. 9.03 Levying and Payment of General Assessments. Based on the duly adopted annual operating budget, the Board of Directors shall levy General Assessments against the Unit Owners in proportion to the percentage interest in the Common Elements appurtenant to each Unit as determined under Section 5.01 of the Declaration. On or before the last day of December of each year, the secretary shall mail or deliver a copy of the annual operating budget and a statement of assessment for the next twelve (12) months to each Unit Owner. General Assessments shall be payable to the Association in twelve (12) equal installments which shall be due monthly in advance on the first day of each month. Such installments shall be mailed or delivered to the principal office of the Association and shall be deemed paid on the date of mailing or on the date of delivery, as the case may be.

9.04 <u>Special Assessments</u>. Special Assessments may from time to time be levied against Unit Owners by the Board of Directors for any of the purposes enumerated in the Declaration and shall be due and payable in the manner and upon the date or dates designated by the Board of Directors.

9.05 <u>Association Remedies Upon Nonpayment of Assessments</u>. Any General Assessment or Special Assessment not paid within ten (10) days of the date on which it is due shall bear interest from the day following such due date at the rate of eighteen percent (18%) per year or the highest rate permitted by law, whichever is less. The Association may seek to collect any assessments not paid when due by filing statements of condominium lien against the Units on which they are assessed, by enforcing and foreclosing such liens or by bringing an action for money damages against the Unit Owners personally obligated to pay the delinquent assessments. A suit to recover a money judgment for unpaid assessments shall be maintainable without foreclosing or waiving any lien securing the same. No Unit Owner may waive or otherwise escape liability for the assessments provided herein by nonuse of the Common Elements or abandonment of its Unit.

9.06 <u>Annual Report</u>. Each January, the Board of Directors shall, by formal action, approve a full and clear annual report of all business transacted by the Association during the previous fiscal year, including a report of the Common Expenses, surpluses and assessments collected from each Unit Owner during the year. Copies of the annual report for the previous year shall be mailed or delivered to each Unit Owner at the address in the Association's membership list prior to the third Thursday in February.

ARTICLE X

USE

Each Unit shall be used only for purposes permitted under the Declaration, the Articles, these Bylaws, and any rules and regulations of the Association.

ARTICLE XI

ENFORCEMENT OF CONDOMINIUM DOCUMENTS

It shall be the responsibility of each Unit Owner to see that the occupants and tenants of the Unit owned by such Unit Owner, and the employees, agents, representatives, invitees and guests of such Unit Owner, occupants and tenants, abide by the provisions of the Declaration, Bylaws, Condominium Ownership Act, all rules and regulations of the Association, and any decisions made by the Association, the Board of Directors or any committees of the Association that are authorized by any of the foregoing. Unit Owners should report infractions to the Board of Directors in writing, and the Board of Directors shall reply to the reporting Unit Owner within thirty (30) days concerning the action taken. In the event of a violation of any provision of the Declaration, the Bylaws, the Condominium Ownership Act, any rule or regulation of the Association, or any authorized decision of the Association, the Board of Directors or any committee of the Association, the Board of Directors shall notify the alleged offender. If the violation is not corrected within a reasonable time, the Association may take such action as it deems appropriate, including legal action against the offending Unit Owner or the Unit Owners of the Unit in which such offender is a tenant, occupant, employee, agent, representative, invitee or guest, to correct the violation. In any such action brought against any Unit Owner in which the Association is the prevailing party, the Unit Owner defendant in such action shall pay the Association's costs and actual attorneys' fees. In the event the Association fails to take appropriate enforcement action within thirty (30) days of the Association's receipt of the report of the infraction, any Unit Owner may take appropriate legal action to enforce the provisions of the Declaration, the Bylaws, the Condominium Ownership Act, the rules and regulations of the Association, and any authorized decision of the Association, the Board of Directors, or any committee of the Association.

ARTICLE XII

LIABILITY AND INDEMNITY

12.01 <u>General Scope and Definitions</u>.

(a) The rights of directors and officers of the Association provided in this Article shall extend to the fullest extent permitted by the Wisconsin Nonstock Corporation Law and other applicable laws as in effect from time to time.

(b) For purposes of this Article, "director or officer" means a natural person (i) who is or was a director or officer of the Association or (ii) who, while a director or officer of the Association, is or was serving at the Association's request as a director, officer, partner, trustee, member of any governing or decisionmaking committee, employee or agent of another corporation or foreign corporation, partnership, limited liability company, joint venture, trust or other enterprise or (iii) who, while a director or officer of the Association, is or was serving an employee benefit plan because his or her duties to the Association also imposed duties on, or otherwise involved services by, the person to the plan or to participants in or beneficiaries of the plan or (iv) who is or was a member of the Architectural Review Committee. Unless the context requires otherwise, "director or officer" shall also mean the estate and personal representative of a director or officer.

(c) For purposes of this Article, "proceeding" means any threatened, pending or completed civil, criminal, administrative or investigative action, suit, arbitration or other proceeding, whether formal or informal, which involves foreign, federal, state or local law (including federal or state securities laws) and which is brought by or in the right of the Association or by any other person.

(d) For purposes of this Article, "expenses" means fees, costs, charges, disbursements, attorneys' fees and any other expenses incurred in connection with a proceeding, including a proceeding in which a director or officer asserts his or her rights under this Article, and, if the context requires, liabilities, including the obligation to pay a judgment, settlement, penalty, assessment, forfeiture or fine, including any excise tax assessed with respect to an employee benefit plan.

12.02 Mandatory Indemnification.

(a) To the extent that a director or officer has been successful on the merits or otherwise in the defense of any proceeding (including, without limitation, the settlement, dismissal, abandonment or withdrawal of any action where he or she does not pay or assume any material liability), or in connection with any claim, issue or matter therein, he or she shall be indemnified by the Association against expenses actually and reasonably incurred by him or her in connection therewith to the extent that he or she was a party to the proceeding because he or she is or was a director or officer of the Association.

In cases not included under Section 12.02(a), the Association (b) shall indemnify any director or officer against expenses actually and reasonably incurred by the director or officer in a proceeding to which the director or officer was a party because he or she is or was a director or officer, unless liability was incurred because the director or officer breached or failed to perform a duty he or she owed to the Association and the breach or failure to perform constituted any of the following: (i) a willful failure to deal fairly with the Association or its members in connection with a matter in which the director or officer had a material conflict of interest; (ii) a violation of criminal law, unless the director or officer had reasonable cause to believe his or her conduct was lawful or no reasonable cause to believe his or her conduct was unlawful; (iii) a transaction from which the director or officer derived an improper personal profit or benefit; or (iv) willful misconduct. The termination of a proceeding by judgment, order, settlement or conviction, or upon a plea of no contest or an equivalent plea, does not, by itself, create a presumption that indemnification of the director or officer is not required under this subsection.

(c) Indemnification under this Section is not required to the extent that the director or officer has previously received indemnification or allowance of expenses from any person, including the Association, in connection with the same proceeding.

(d) To the extent indemnification is required under this Article XI, the Association has purchased or is required under Section 12.10 to purchase insurance on behalf of the indemnified person and the insurance policy includes a provision obligating the insurer to defend such person, the Association shall be obligated to extend such defense. To the extent possible under such insurance policy, the defense shall be extended with counsel reasonably acceptable to the indemnified person. The Association shall keep the indemnified person advised of the status of the claim and the defense thereof and shall consider in good faith the recommendations made by the indemnified person with respect thereto.

12.03 Determination of Right to Indemnification. Unless otherwise provided by written agreement between the director or officer and the Association, the director or officer seeking indemnification under Section 12.02 shall make a written request for indemnification which shall designate one of the following means for determining his or her right to indemnification: (a) by a majority vote of a quorum of the Board of Directors or a committee of directors consisting of directors not at the time parties to the same or related proceedings; (b) by independent legal counsel selected by a quorum of the Board of Directors or its committee in the manner prescribed in Section 12.03(a) or, if unable to obtain such a quorum or committee, by a majority vote of the full Board of Directors, including directors who are parties to the same or related proceedings; (c) by arbitration; or (d) by an affirmative vote of a majority of the Unit Owners entitled to vote; provided, however, that Unit Owners who are at the time parties to the same or related proceedings, whether as plaintiffs or defendants or in any other capacity, may not vote in making the determination. Any determination under this Section shall be made pursuant to procedures consistent with the Wisconsin Nonstock Corporation Law unless otherwise agreed by the Association and the person seeking indemnification. Such determination shall be completed, and eligible expenses, if any, shall be paid to the person requesting indemnification hereunder within sixty (60) days of the Association's receipt of the written request required hereunder.

12.04 <u>Allowance of Expenses as Incurred</u>. Within thirty (30) days after a written request by a director or officer who is a party to a proceeding because he or she is or was a director or officer, the Association shall pay or reimburse his or her reasonable expenses as incurred if the director or officer provides the Association with all of the following: (a) a written affirmation of his or her good faith belief that he or she has not breached or failed to perform his or her duties to the Association; and (b) a written undertaking, executed personally or on his or her behalf, to repay the allowance and, if required by the Association, to pay reasonable interest on the allowance to the extent that it is ultimately determined under Section 12.03 that indemnification under Section 12.02 is not required and indemnification is otherwise not ordered by a court. The undertaking under this Section shall be an unlimited general obligation of the director or officer and may be accepted without reference to his or her ability to repay the allowance. The undertaking may be secured or unsecured.

12.05 Partial Indemnification.

(a) If it is determined pursuant to Section 12.03 that a director or officer is entitled to indemnification as to some claims, issues or matters in connection with any proceeding, but not as to other claims, issues or matters, the person or persons making such determination shall reasonably determine and indemnify the director or officer for those expenses which are the result of claims, issues or matters that are a proper subject for indemnification hereunder in light of all of the circumstances.

(b) If it is determined pursuant to Section 12.03 that certain expenses (other than liabilities) incurred by a director or officer are for any reason unreasonable in amount in light of all the circumstances, the person or persons making such determination shall authorize the indemnification of the director or officer for only such amounts as he or she or they shall deem reasonable.

12.06 <u>Indemnification of Employees and Agents</u>. The Board of Directors, may, in its sole discretion, provide indemnification and/or defense and/ or allowance of expenses in advance of a final determination of any proceeding to an employee or agent of the Association who is not a director or officer in connection with any proceeding in which the employee or agent was a defendant because of his or her actions as an employee or agent of the Association; provided, however, that prior to such indemnification, defense or allowance of expenses, the Board of Directors shall first determine that the employee or agent acted in good faith and in a manner he or she reasonably believed to be in, and not opposed to, the best interests of Association.

12.07 Limited Liability of Directors and Officers.

(a) Except as provided in Sections 12.07(b) and 12.07(c), a director or officer is not liable to the Association, its members or creditors, or any person for damages, settlements, fees, fines, penalties or other monetary liabilities arising from a breach of, or failure to perform, any duty resulting solely from his or her status as a director or officer, unless the person asserting liability proves that the breach or failure to perform constitutes any of the acts of misconduct listed in Section 12.02(b).

(b) Except as provided in Section 12.07(c), this Section 12.07 does not apply to any of the following: (i) a civil or criminal proceeding brought by or on behalf of any governmental unit, authority or agency;(ii) a proceeding brought by any person for a violation of state or federal

law where the proceeding is brought pursuant to an express private right of action created by state or federal statute; or (iii) the liability of a director under Wisconsin Statutes Sections 181.0832 and 181.0833.

(c) Sections 12.07(b)(i) and (ii) do not apply to a proceeding brought by a governmental unit, authority or agency in its capacity as a private party or contractor.

12.08 <u>Severability of Provisions</u>. The provisions of this Article 12.08 and the several rights to indemnification, advancement of expenses and limitation of liability created hereby are independent and severable and, in the event that any such provision or right shall be held by a court of competent jurisdiction in which a proceeding relating to such provisions or rights is brought to be against public policy or otherwise to be unenforceable, the other provisions of this Article 12.08 shall remain enforceable and in full effect.

12.09 Nonexclusivity of Rights. The rights to indemnification, defense and advancement of expenses provided for in this Article shall not be deemed exclusive of any other rights to which those seeking indemnification, defense or advancement of expenses may be entitled under any agreement authorized by the Board of Directors, any of the Bylaws, any vote of the members or disinterested directors or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding such office. Notwithstanding the foregoing, the Association may not indemnify a director or officer, or permit a director or officer to retain any allowance of expenses, pursuant to any such additional rights unless it is determined by or on behalf of the Association that the director or officer did not breach or fail to perform a duty he or she owes to the Association which constitutes conduct under Section 12.02(b). A director or officer who is a party to the same or related proceeding for which indemnification, defense or an allowance of expenses is sought may not participate in a determination under this Section.

12.10 <u>Purchase of Insurance</u>. The Association shall use its reasonable best efforts to purchase and maintain insurance on behalf of any person who is or was a director or officer of the Association, to the extent that such director or officer is insurable and such insurance coverage can be secured by the Association at rates, and in amounts and subject to such terms and conditions as shall be determined in good faith to be reasonable and appropriate by the Board of Directors of the Association, and whose determination shall be conclusive (provided, however, that such insurance shall contain a provision obligating the insurer to defend the director or officer, if such provision is available at reasonable rates), against liability asserted against or incurred by him or her in any such capacity or arising out of his or her status as such, whether or not the Association would have the power to indemnify or defend him or her against such liability under the provisions of this Article.

12.11 <u>Benefit</u>. The rights to indemnification, defense and advancement of expenses provided by, or granted pursuant to, this Article shall continue as to a person who has ceased to be a director or officer and shall inure to the benefit of the heirs, executors and administrators of such a person.

12.12 <u>Amendment</u>. No amendment or repeal of this Article shall be effective to reduce the obligations of the Association under this Article with respect to any proceeding based upon occurrences which take place prior to such amendment or repeal.

ARTICLE XIII

GENERAL PROVISIONS

13.01 <u>Seal</u>. The Association shall not have a corporate seal.

13.02 <u>Interpretation</u>. These Bylaws are subject to all provisions of the Declaration, the Articles, the Wisconsin Condominium Ownership Act and the Wisconsin Nonstock Corporation Law. In the event any provision of these Bylaws shall be held invalid, such invalidity shall not render invalid any other provision hereof which can be given effect. Any invalid provision or portion thereof shall be interpreted as having been amended to comply with the provisions of the Wisconsin Condominium Ownership Act and/or the Wisconsin Nonstock Corporation Law in effect on the date of the adoption of these Bylaws. Nothing in these Bylaws shall be deemed or construed to authorize the Association to conduct or engage in any active business for profit on behalf of any or all of the Unit Owners.

13.03 <u>Notices</u>. Except as otherwise may be provided in the Wisconsin Condominium Ownership Act or Wisconsin Nonstock Corporation Law, notices to any Unit Owner that are to be delivered or mailed pursuant to these Bylaws shall be deemed to have been given (a) in the case of delivered notices, on the date when the notice is delivered to the address on file with the secretary of the Association, or (b) in the case of mailed notices, on the date when the notice, addressed to the address on file with the secretary of the Association, is deposited in the United States mail with sufficient postage to effect delivery.

ARTICLE XIV

AMENDMENT

These Bylaws may be amended only with the assent of at least sixtyseven percent (67%) of the votes of the Unit Owners; provided, however, as long as the Declarant owns any Unit, no amendment shall be effective without the written consent of the Declarant. Any first Mortgagee or its insurer or guarantor shall, upon written request to the Association, be entitled to timely written advance notice of any proposed amendment to these Bylaws.

RULES AND REGULATIONS OF THE ASSOCIATION

The following rules and regulations are adopted by BESA UNIT OWNERS ASSOCIATION, INC. (the "Association") for the purpose of assuring that the Condominium is operated in an efficient and orderly manner so as to create a pleasant living environment.

ARTICLE I

GENERAL

1.01 <u>Applicability to All Residents</u>. All rules and regulations shall apply to and shall be complied with by all Unit Owners, residents within Units and their guests, families, invitees and tenants.

1.02 <u>Definitions</u>. All capitalized terms not defined herein shall have the definitions assigned to such terms by the Declaration of Condominium for Barracuda Condominium (the "Declaration").

1.03 <u>Keys and Locks</u>. The Association shall have the right to retain a passkey to each Unit at all times for the event of emergencies. No Unit Owner shall alter any lock, or install a new lock on any door of the Condominium without the prior written consent of the Association. If such consent is given, the Unit Owner shall provide the Association with an additional key for use by the Association pursuant to its rights to access the Units.

1.04 <u>Winter Heating</u>. Whether occupied or vacant, all Units shall be heated to at least 60° Fahrenheit during the winter months.

ARTICLE II

APPEARANCE

2.01 <u>Signs</u>. Commercial and Live/Work Unit Owners are allowed One (1) exterior sign complying with and approved by, the local Authority having jurisdiction for size, illumination, location, and permitting. Other signs of any kind shall not be displayed to the public view on any Unit without the written consent of the Association except for political signs and, if Declarant owns at least one Unit, the Declarant. The Declarant reserves the right to erect signs, gates or other entryway features at all entrances to the Condominium and to erect appropriate signs for the sale of Units.

2.02 <u>Protrusions</u>. No awning, machines, air conditioning units, wiring for electrical or telephone installation or other similar protrusions shall be allowed on the exterior of the Condominium without the prior written consent of the Association.

2.03 <u>Antennae</u>. To the extent this restriction is permitted by applicable law, no exterior antennas, windmills or satellite dishes shall be erected on any Unit or on any Common Elements without the prior written approval of the Association.

2.04 <u>Laundry</u>. No laundry is to be hung on the decks, patios or in windows for any reason.

2.05 <u>Limited Common Elements</u>. All decks and patios which are open to public view shall be kept in a neat and orderly condition. No personal property shall be stored thereon except for hard piped natural gas grills and for patio and deck furniture. Unit Owners may display holiday decorations and political signs.

ARTICLE III

USE RESTRICTIONS

3.01 <u>Animals</u>. Unit Owners shall be allowed to keep up to: (i) two (2) cats or one (1) dog per Unit; or (ii) one (1) dog and one (1) cat per Unit; or (iii) subject to a combined weight limit of one hundred (100) pounds, two (2) dogs per unit. Animals shall be carried or kept on a leash at all times when not in the Units. In addition to cats and dogs, small animals that are kept in a cage or tank may also be permitted with Association approval. All animals shall not unreasonably disturb other Condominium residents and shall not be left unattended in any portion of the Common Elements. All animals must be registered with the Association and owners of animals shall be pecuniarily liable for damage caused by their animals. Kennels shall be kept inside their respective owner's Unit. Unit Owners are responsible for the immediate clean-up of their animals regardless of the circumstances.

3.02 <u>Damage to Common Elements</u>. Damages to the Common Elements caused by a resident or visitors of a resident or an agent of a resident shall be the responsibility of the Unit Owner or the person causing such damage.

3.03 <u>Maintenance of Unit</u>. All Unit Owners shall promptly perform or shall have promptly performed all maintenance and repair work within their own Unit which would adversely affect any portion of the Condominium. Each Unit Owner shall be responsible for all damages and liabilities that any failure to maintain or repair may engender.

2

3.04 <u>Maintenance of Common Elements</u>. Unit Owners shall be prohibited from discarding any materials from the windows, balconies or doors of the Units and shall be prohibited from discarding any materials into the Common Elements.

3.05 <u>Nuisances</u>. No offensive or unlawful activity shall occur in the Condominium. No offensive or unlawful use shall be made of the Condominium. All Unit Owners at their own expense shall comply with all city, state and federal laws applicable to their Unit. No Unit shall be used or maintained as a dumping ground for garbage.

3.06 <u>Storage</u>. The Association shall not be liable for any loss or damage to property placed in any Unit or Common Elements. No materials, prohibited by law or local ordinance may be stored in any of these areas.

3.07 <u>Salting</u>. Unit Owners are hereby prohibited from using salt on the access walks or driveways of the Condominium and shall be liable for the costs of repairing all damage caused by the use of salt.

3.08 <u>Commercial Units</u>. Commercial and Live/Work Unit Owners shall not use or permit the use of their Units in violation of any local ordinances or zoning restrictions.

ARTICLE IV

VEHICLE RESTRICTIONS

4.01 <u>Obstructions</u>. Driveways shall not be used for any purpose other than the ingress and egress to and from Units.

4.02 <u>Parking</u>. Unit Owners shall not park, nor shall they permit their families, guests, invitees or tenants to park upon or to block access to, the parking areas of other Unit Owners. Improperly parked vehicles shall be subject to removal at vehicle owner's expense. Unit Owners shall not leave their vehicles idling in any garages.

4.03 <u>Service and Recreational Vehicles</u>. Parking of service and recreational vehicles, including but not limited to trailers, boats, campers, vans or other vehicles shall be prohibited unless such vehicles are kept in the Unit Owner's parking area. These provisions shall not prohibit temporary parking of such vehicles for the purpose of loading and unloading. A temporary waiver of these prohibitions may be obtained at the discretion of the Association.

4.04 <u>Vehicle Repairs</u>. No vehicle maintenance or repairs shall be allowed anywhere outside of the Unit.

4.05 <u>Bikes/Recreational Equipment</u>. Unit Owners shall keep bikes and other recreational equipment in their Unit and not stored in the Common Elements except bikes may be stored in those areas designated as bike storage areas on the floorplans.

4.06 <u>Garage Door</u>. The garage door to the underground parking area shall remain closed at all times except when in use for ingress or egress purposes.

ARTICLE V

AMENDMENTS

This document may be amended as provided in Section 6.01(k) of the Bylaws.

ARTICLE VI

FINES

In addition to all other remedies available to the Association or to other Unit Owners under the Declaration, the Bylaws or applicable law, the Association shall have the right, following delivery of notice of violation and expiration of any cure period required under the Declaration, to impose against any Unit Owner in violation of the Declaration, the Bylaws or these Rules and Regulations, to impose a fine against such Unit Owner according to the following schedule:

- (a) For the first offense in a given calendar year: **\$100.00**
- (b) For the second offense in any given calendar year: **\$200.00**
- (c) For the third offense in any given calendar year: **\$500.00**

Fines are to be paid immediately to the Association. Any fine not paid within ten (10) days after billing therefor by the Association shall accrue a late charge in the amount of \$25.00 for every month the fine is not paid. The Association shall have the right, following imposition of any fine, to collect the same as a Special Assessment against the Unit Owner's Unit.